

ATLAS Paper Draft

Search for an additional, heavy Higgs boson in the $H \rightarrow ZZ$ decay channel at $\sqrt{s} = 8$ TeV in pp collision data with the ATLAS detector

HIGG-2013-20

Version: 2.0

To be submitted to: EPJC

Supporting internal notes

$$\begin{split} H &\to ZZ \to \ell^+ \ell^- q\bar{q}: \text{https://cds.cern.ch/record/1693159} \\ H &\to ZZ \to \ell^+ \ell^- v \bar{v}: \text{https://cds.cern.ch/record/1693161} \\ H &\to ZZ \to \ell^+ \ell^- \ell^+ \ell^-: \text{https://cds.cern.ch/record/1693487} \\ H &\to ZZ \to v \bar{v} q \bar{q}: \text{https://cds.cern.ch/record/1692942} \\ \text{Combination: https://cds.cern.ch/record/1995509} \end{split}$$

Comments are due by: 25 June 2015

Abstract

A search is presented for a high-mass Higgs boson in the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- \ell^+ \ell^-$, $H \rightarrow ZZ \rightarrow \ell^+ \ell^- v \bar{v}$, $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q \bar{q}$, and $H \rightarrow ZZ \rightarrow v \bar{v} q \bar{q}$ decay modes using the ATLAS detector at the CERN Large Hadron Collider. The search uses proton-proton collision data at a centre-of-mass energy of 8 TeV corresponding to an integrated luminosity of 20.3 fb⁻¹. The results of the search are interpreted in the scenario of a heavy Higgs boson with a width that is small compared with the experimental mass resolution. The Higgs boson mass range considered extends up to 1 TeV for all four decay modes and down to as low as 140 GeV, depending on the decay mode. No significant excess of events over the Standard Model prediction is found. A simultaneous fit to the four decay modes yields upper limits on the production cross-section of a heavy Higgs boson times the branching ratio to Z boson pairs. 95% confidence level upper limits range from 0.19 pb (at $m_H = 140 \text{ GeV})$ to 0.011 pb (at $m_H = 1 \text{ TeV}$) for the gluon-fusion production mode, and from 0.20 pb (at $m_H = 140 \text{ GeV})$ to 0.013 pb at ($m_H = 1 \text{ TeV}$) for the vector-boson-fusion production mode. The results are also interpreted in the context of Type-I and Type-II two-Higgs-doublet models.

Document created on 30th June 2015 using ATLAS LATEX Version 01-07-04.

Analysis Team

[email: atlas-HIGG-2013-20-editors@cern.ch]

S. H. Abidi, C. Anastopoulos, S. Angelidakis, L. Aperio Bella, M. Baak,
T. Donszelmann Cuhadar, N. Edwards, D. Fassouliotis, G. Garrillo Montoya,
F. Garay Walls, C. Gwilliam, M. Hoffmann, R. Harrington, L.S. Kaplan,
D. Kyriazopoulos, A. Mehta, E. Mountricha, R. Nicolaidou, S. Oda, H. Okawa,
K. Parker, A. Sanchez Pineda, F. Lo Sterzo, S. Snyder, C. Wang, S.M. Wang,
L. Xu, L. Zhang

Editorial Board

[email: atlas-HIGG-2013-20-editorial-board@cern.ch]

M. Vos (chair) V. Cavaliere P. Maettig D. Whiteson



ATLAS PAPER

HIGG-2013-20

30th June 2015



Draft version 2.0

² Search for an additional, heavy Higgs boson in the $H \rightarrow ZZ$ decay ³ channel at $\sqrt{s} = 8$ TeV in *pp* collision data with the ATLAS ⁴ detector

The ATLAS Collaboration

Abstract

A search is presented for a high-mass Higgs boson in the $H \to ZZ \to \ell^+ \ell^- \ell^+ \ell^-$, $H \to ZZ \to \ell^+ \ell^- \ell^+ \ell^-$, $H \to ZZ \to \ell^+ \ell^- \ell^+ \ell^-$ 7 $ZZ \rightarrow \ell^+ \ell^- v \bar{v}, H \rightarrow ZZ \rightarrow \ell^+ \ell^- q \bar{q}, \text{ and } H \rightarrow ZZ \rightarrow v \bar{v} q \bar{q} \text{ decay modes using the ATLAS}$ 8 detector at the CERN Large Hadron Collider. The search uses proton-proton collision data 9 at a centre-of-mass energy of 8 TeV corresponding to an integrated luminosity of 20.3 fb^{-1} . 10 The results of the search are interpreted in the scenario of a heavy Higgs boson with a width 11 that is small compared with the experimental mass resolution. The Higgs boson mass range 12 considered extends up to 1 TeV for all four decay modes and down to as low as 140 GeV, 13 depending on the decay mode. No significant excess of events over the Standard Model 14 prediction is found. A simultaneous fit to the four decay modes yields upper limits on the 15 production cross-section of a heavy Higgs boson times the branching ratio to Z boson pairs. 16 95% confidence level upper limits range from 0.19 pb (at $m_H = 140$ GeV) to 0.011 pb (at 17 $m_H = 1$ TeV) for the gluon-fusion production mode, and from 0.20 pb (at $m_H = 140$ GeV) 18 to 0.013 pb at $(m_H = 1 \text{ TeV})$ for the vector-boson-fusion production mode. The results are 19 also interpreted in the context of Type-I and Type-II two-Higgs-doublet models. 20

To be submitted to: EPJC

© 2015 CERN for the benefit of the ATLAS Collaboration.

²³ Reproduction of this article or parts of it is allowed as specified in the CC-BY-3.0 license.

21 22

5

6

24 Contents	
-------------	--

25	1	Introduction	3
26	2	ATLAS detector	4
27	3	Data and Monte Carlo samples	4
28		3.1 Data sample	4
29		3.2 Signal samples and modelling	5
30		3.3 Background samples	6
31		3.4 Detector simulation	7
32	4	Object reconstruction and common event selection	7
33	5	$H \rightarrow ZZ \rightarrow \ell^+ \ell^- \ell^+ \ell^-$ event selection and background estimation	10
34		5.1 Event selection	10
35		5.2 Background estimation	11
36	6	$H \rightarrow ZZ \rightarrow \ell^+ \ell^- v \bar{v}$ event selection and background estimation	13
37		6.1 Event selection	13
38		6.2 Background estimation	14
39	7	$H \rightarrow ZZ \rightarrow \ell^+ \ell^- q \bar{q}$ event selection and background estimation	16
40		7.1 Event selection	16
41		7.1.1 Resolved ggF channel	16
42		7.1.2 Merged-jet ggF channel	17
43		7.1.3 VBF channel	18
44		7.2 Background estimation	19
45	8	$H \rightarrow ZZ \rightarrow v \bar{v} q \bar{q}$ event selection and background estimation	23
46		8.1 Event selection	23
47		8.2 Background estimation	24
48	9	Systematic uncertainties	26
49		9.1 Experimental uncertainties	27
50		9.2 Signal acceptance uncertainty	28
51		9.3 $ZZ^{(*)}$ background uncertainties	28
52	10	Combination and statistical interpretation	29
53	11	Results	32
54	12	Summary	36
55	A	Corrections to MC simulation for the $\ell\ell q q$ search	37
56	B	Flavour tagging in the $\ell \ell q q$ and $\nu \nu q q$ search	37

57 1. Introduction

⁵⁸ In 2012, a Higgs boson *h* with a mass of 125 GeV was discovered by the ATLAS and CMS collaborations ⁵⁹ at the LHC [1, 2]. One of the most important remaining questions is whether the newly discovered particle ⁶⁰ is part of an extended scalar sector as postulated by various extensions to the Standard Model (SM) such ⁶¹ as the two-Higgs-doublet model (2HDM) [3] and the electroweak-singlet (EWS) model [4]. These predict ⁶² additional Higgs bosons, motivating searches at masses other than 125 GeV.

 $_{63}$ This paper reports four separate searches with the ATLAS detector for a heavy neutral scalar H boson

decaying into two SM Z bosons, encompassing the decay modes $ZZ \rightarrow \ell^+ \ell^- \ell^+ \ell^-$, $ZZ \rightarrow \ell^+ \ell^- v \bar{v}$, $ZZ \rightarrow \ell^+ \ell^- q \bar{q}$, and $ZZ \rightarrow v \bar{v} q \bar{q}$, where ℓ stands for either an electron or a muon. These modes are

referred to, respectively, as $\ell\ell\ell\ell$, $\ell\ell\nu\nu$, $\ell\ell qq$, and $\nu\nu qq$. Some of these searches are further divided into

channels based on the production mode (see Table 3).

It is assumed that additional Higgs bosons would be produced predominantly via the gluon fusion (ggF) 68 and vector-boson-fusion (VBF) processes but that the ratio of the two production mechanisms is unknown 69 in the absence of a specific model. For Higgs boson masses below 200 GeV, associated production (VH, 70 where V stands for either a W or a Z boson) is important as well. In this mass range, only the $\ell\ell\ell\ell$ decay 71 mode is considered. Due to its excellent mass resolution and high signal-to-background ratio, the $\ell\ell\ell\ell\ell$ 72 decay mode is well-suited for a search for a narrow resonance in the range $140 < m_H < 500$ GeV. The 73 $\ell\ell\ell\ell\ell$ search includes channels sensitive to VH production as well as VBF and ggF production modes. The 74 $\ell\ell qq$ and $\ell\ell\nu\nu$ searches, which start at $m_H = 200$ GeVa nd $m_H = 240$ GeV respectively, have ggF and 75 VBF channels only. The $\nu \nu qq$ search starts at $m_H = 400$ GeV and has channels inclusive in ggF and VBF. 76 When results from all four searches are combined, the $\ell \ell q q$, $\ell \ell \nu \nu$ and $\nu \nu q q$ decay modes contribute to the 77 overall sensitivity, especially in the VBF production mode, and become dominant for $m_H > 500$ GeV due 78 to their larger branching fractions. All four searches are done for Higgs boson masses up to 1000 GeV. 79

Searches are further divided into channels based on characteristics of the final state such as lepton flavours and the possible presence of heavy-flavour jets. The ggF search for the $\ell\ell\ell\ell\ell$ decay mode is divided into 4 channels based on lepton flavour, while for $\ell\ell\nu\nu$ both ggF and VBF channels are divided into 2 channels each based on lepton flavour. For the $\ell\ell qq$ and $\nu\nu qq$ decay modes, the ggF searches are divided into two channels each based on the number of *b*-tagged jets in the event. For Higgs boson masses above 700 GeV, jets from *Z* decay are boosted and tend to merge into single jets, so the $\ell\ell qq$ search has an additional channel designed to search for Higgs events with merged jets.

For each channel, a discriminating variable sensitive to m_H is identified and used in a likelihood fit. 87 The $\ell\ell\ell\ell$ and $\ell\ell qq$ searches use the invariant mass of the four-fermion system as the final discriminant, 88 while the $\ell\ell vv$ and vvqq searches use a transverse mass distribution. Distributions of these discriminants 89 for each channel are combined in a simultaneous likelihood fit which estimates the rate of heavy Higgs 90 boson production and simultaneously the nuisance parameters corresponding to systematic uncertainties. 91 Additional distributions from background-dominated control regions also enter the fit in order to constrain 92 nuisance parameters. Unless otherwise stated, all figures show shapes and normalizations determined 93 from this fit. All results are interpreted in the scenario of a new Higgs boson with a narrow width (the 94 narrow-width approximation), as well as in Type-I and Type-II 2HDMs. 95

⁹⁷ in the $\ell\ell\ell\ell$, $\ell\ell qq$, and $\ell\ell\nu\nu$ modes with 4.7–4.8 fb⁻¹ of data collected at $\sqrt{s} = 7$ TeV [5–7]. A heavy

Higgs boson with the width and branching fractions predicted by the SM was excluded at the 95% confidence level in the ranges $182 < m_H < 233$ GeV, $256 < m_H < 265$ GeV, and $268 < m_H < 415$ GeV

by the $\ell\ell\ell\ell\ell$ mode; in the ranges $300 < m_H < 322$ GeV and $353 < m_H < 410$ GeV by the $\ell\ell qq$ mode; and in the range $319 < m_H < 558$ GeV by the $\ell\ell\nu\nu$ mode. The searches in this paper improve on the earlier results by using a larger data set of 20.3 fb⁻¹ of *pp* collision data collected at a higher centre-of-mass energy of $\sqrt{s} = 8$ TeV, by adding the $\nu\nu qq$ decay mode, by further optimizing the event selection and other aspects of the analysis, and by combining results of all four searches. The CMS Collaboration has also recently published a search for a heavy Higgs boson in $H \rightarrow ZZ$ decays [8].

This paper is organized as follows. After a brief description of the ATLAS detector in Section 2, the simulation of the background and signal processes used in this analysis is outlined in Section 3. Section 4 summarizes the reconstruction of the final-state objects used by these searches. The event selection and background estimation for the four searches are presented in Sections 5 to 8, and Section 9 discusses the systematic uncertainties common to all searches. Section 10 details the statistical combination of all the searches into a single limit, which is given in Section 11. Finally, Section 12 gives the conclusions.

112 2. ATLAS detector

ATLAS is a multi-purpose detector [9] which provides nearly full solid-angle coverage around the inter-113 action point.¹ It consists of a tracking system (inner detector or ID) surrounded by a thin superconducting 114 solenoid providing a 2 T magnetic field, electromagnetic and hadronic calorimeters, and a muon spectro-115 meter (MS). The ID consists of pixel and silicon microstrip detectors covering the pseudorapidity region 116 $|\eta| < 2.5$, surrounded by a transition radiation tracker (TRT), which improves electron identification 117 in the region $|\eta| < 2.0$. The sampling calorimeters cover the region $|\eta| < 4.9$. The forward region 118 $(3.2 < |\eta| < 4.9)$ is instrumented with a liquid-argon (LAr) calorimeter for electromagnetic and hadronic 119 measurements. In the central region, a high-granularity lead/LAr electromagnetic calorimeter covers 120 $|\eta| < 3.2$. Hadron calorimetry is based on either steel absorbers with scintillator tiles ($|\eta| < 1.7$) or 121 copper absorbers in LAr (1.5 < $|\eta|$ < 3.2). The MS consists of three large superconducting toroids 122 arranged with an eight-fold azimuthal coil symmetry around the calorimeters, and a system of three layers 123 of precision gas chambers providing tracking coverage in the range $|\eta| < 2.7$, while dedicated chambers 124 allow triggering on muons in the region $|\eta| < 2.4$. The ATLAS trigger system [10] consists of three levels; 125 the first (L1) is a hardware-based system, while the second and third levels are software-based systems. 126

3. Data and Monte Carlo samples

128 **3.1. Data sample**

The data used in these searches were collected by ATLAS at a centre-of-mass energy of 8 TeV during 2012 and correspond to an integrated luminosity of 20.3 fb^{-1} .

¹ ATLAS uses a right-handed coordinate system with its origin at the nominal interaction point (IP) in the centre of the detector and the *z*-axis coinciding with the axis of the beam pipe. The *x*-axis points from the IP towards the centre of the LHC ring, and the *y*-axis points upward. Cylindrical coordinates (r,ϕ) are used in the transverse plane, ϕ being the azimuthal angle around the *z*-axis. The pseudorapidity is defined in terms of the polar angle θ as $\eta = -\ln \tan(\theta/2)$. The distance in (η,ϕ) coordinates, $\Delta R = \sqrt{(\Delta\phi)^2 + (\Delta\eta)^2}$, is also used to define cone sizes. Transverse momentum and energy are defined as $p_T = p \sin \theta$ and $E_T = E \sin \theta$, respectively.

Collision events are recorded only if they are selected by the online trigger system. For the vvqq search this 131 selection requires that the magnitude $E_{\rm T}^{\rm miss}$ of the missing transverse momentum vector (see Section 4) is 132 above 80 GeV. Searches with leptonic final states use a combination of single-lepton and dilepton triggers 133 in order to maximize acceptance. The main single-lepton triggers have a minimum $p_{\rm T}$ (muons) or $E_{\rm T}$ 134 (electrons) threshold of 24 GeV and require that the leptons are isolated. They are complemented with 135 triggers with higher thresholds (60 GeV for electrons and 36 GeV for muons) and no isolation requirement 136 in order to increase acceptance at high $p_{\rm T}$ and $E_{\rm T}$. The dilepton triggers require two same-flavour leptons 137 with a threshold of 12 GeV for electrons and 13 GeV for muons. The acceptance in the $\ell\ell\ell\ell\ell$ search is 138 increased further with an additional asymmetric dimuon trigger selecting one muon with $p_{\rm T} > 18 {\rm ~GeV}$ 139 and another one with $p_{\rm T} > 8$ GeV and an electron-muon trigger with thresholds of $E_{\rm T}^e > 12$ GeV and 140 $p_{\rm T}^{\mu} > 8 \,{\rm GeV}.$ 141

142 **3.2.** Signal samples and modelling

The acceptance and resolution for the signal of a narrow-width heavy Higgs boson decaying to a Z boson 143 pair are modelled using Monte Carlo (MC) simulation. Signal samples are generated using POWHEG 144 r1508 [11, 12], which calculates separately the gluon and vector-boson-fusion Higgs boson production 145 processes up to next-to-leading order (NLO) in $\alpha_{\rm S}$. The generated signal events are hadronized with 146 PYTHIA 8.165 using the AU2 set of tunable parameters for the underlying event [13, 14]; PYTHIA also 147 decays the Z bosons into all modes considered in this search. The contribution from Z boson decay to 148 τ leptons is also included. The NLO CT10 [15] parton distribution function (PDF) is used. The associated 149 production of Higgs bosons with a W or Z boson (WH and ZH) is significant for $m_H < 200$ GeV. It 150 is therefore included as a signal process for the $\ell\ell\ell\ell\ell$ search for $m_H < 400$ GeV and simulated using 151 PYTHIA 8 with the LO CTEQ6L1 PDF set [16] and the AU2 parameter set. These samples are summarized 152 in Table 1. 153

Besides model-independent results, a search in the the context of a CP-conserving 2HDM [3] is also 154 presented. This model has five physical Higgs bosons after electroweak symmetry breaking: two CP-155 even, h and H; one CP-odd, A; and two charged, H^{\pm} . The model considered here has seven free 156 parameters: the Higgs boson masses $(m_h, m_H, m_A, m_{H^{\pm}})$, the ratio of the vacuum expectation values 157 of the two doublets $(\tan \beta)$, the mixing angle between the CP-even Higgs bosons (α) , and the potential 158 parameter m_{12}^2 that mixes the two Higgs doublets. The two Higgs doublets Φ_1 and Φ_2 can couple to 159 leptons and up- and down-type quarks in several ways. In the Type-I model, Φ_2 couples to all quarks 160 and leptons, whereas for Type-II, Φ_1 couples to down-type quarks and leptons and Φ_2 couples to up-type 161 quarks. The 'lepton-specific' model is similar to Type-I except for the fact that the leptons couple to Φ_1 , 162 instead of Φ_2 ; the 'flipped' model is similar to Type-II except that the leptons couple to Φ_2 , instead of 163 Φ_1 . In all these models, the coupling of the H boson to vector bosons is proportional to $\cos(\beta - \alpha)$. In 164 the limit $\cos(\beta - \alpha) \rightarrow 0$ the light CP-even Higgs boson, h, is indistinguishable from a SM Higgs boson 165 with the same mass. In the context of $H \rightarrow ZZ$ decays there is no direct coupling of the Higgs boson to 166 leptons, and so only the Type-I and -II interpretations are presented. 167

¹⁶⁸ The production cross-sections for both the ggF and VBF processes are calculated using SusHi 1.3.0 [17–

¹⁶⁹ 22], while the branching ratios are calculated with 2HDMC 1.6.4 [23]. For the branching ratio calculations

it is assumed that $m_A = m_H = m_{H^{\pm}}$, $m_h = 125$ GeV, and $m_{12}^2 = m_A^2 \tan \beta / (1 + \tan \beta^2)$. In the 2HDM

parameter space considered in this analysis, the cross-section times branching ratio for $H \rightarrow ZZ$ with

 $m_H = 200 \text{ GeV}$ varies from 2.4 fb to 10 pb for Type-I and from 0.5 fb to 9.4 pb for Type-II.

The width of the heavy Higgs boson varies over the parameter space of the 2HDM model, and may be 173 significant compared with the experimental resolution. Since this analysis assumes a narrow-width signal, 174 the 2HDM interpretation is limited to regions of parameter space where the width is less than 0.5% of m_H 175 (significantly smaller than the detector resolution). In addition, the off-shell contribution from the light 176 Higgs boson and its interference with the non-resonant ZZ background vary over the 2HDM parameter 177 space as the light Higgs boson couplings are modified from their SM values. Therefore the interpretation 178 is further limited to regions of the parameter space where the light Higgs boson couplings are enhanced by 179 less than a factor of three from their SM values; in these regions the variation is found to have a negligible 180 effect. 181

182 3.3. Background samples

Monte Carlo simulations are also used to model the shapes of distributions from many of the sources of SM background to these searches. Table 1 summarizes the simulated event samples along with the PDF sets and underlying-event tunes used. Additional samples are also used to compute systematic uncertainties as detailed in Section 9.

SHERPA 1.4.1 [24] includes the effects of heavy-quark masses in its modelling of the production of W and 187 Z bosons along with additional jets (V + jets). For this reason it is used to model these backgrounds 188 in the hadronic $\ell \ell q q$ and $\nu \nu q q$ searches, which are subdivided based on whether the Z boson decays 189 into b-quarks or light-flavour quarks. The ALPGEN 2.14 W + jets and Z/γ^* + jets samples are generated 190 with up to five hard partons and with the partons matched to final-state particle jets [25, 26]. They are 191 used to describe these backgrounds in the other decay modes and also in the VBF channel of the $\ell \ell q q$ 192 search² since the additional partons in the matrix element give a better description of the VBF topology. 193 The SHERPA (ALPGEN) Z/γ^* + jets samples have a dilepton invariant mass requirement of $m_{\ell\ell} > 40$ GeV 194 (60 GeV) at the generator level. 195

The background from the associated production of the 125 GeV *h* boson along with a *Z* boson is nonnegligible in the $\ell\ell qq$ and $\nu\nu qq$ searches and is taken into account. Contributions to *Zh* from both $q\bar{q}$ annihilation and gluon fusion are included. The $q\bar{q} \rightarrow Zh$ samples take into account NLO electroweak corrections, including differential corrections as a function of *Z* boson $p_{\rm T}$ [27, 28]. The Higgs boson branching ratio is calculated using HDECAY [29]. Further details can be found in Ref. [30].

Continuum $ZZ^{(*)}$ events form the dominant background for the $\ell\ell\ell\ell$ and $\ell\ell\nu\nu$ decay modes; this is 201 modelled with a dedicated $q\bar{q} \rightarrow ZZ^{(*)}$ sample. This sample is corrected to match the calculation 202 described in Ref. [31], which is next-to-next-to-leading order (NNLO) in $\alpha_{\rm S}$, with a K-factor that is 203 differential in m_{ZZ} . Higher-order electroweak effects are included following the calculation reported in 204 Refs. [32, 33] by applying a K-factor based on the kinematics of the diboson system and the initial-state 205 quarks, using a procedure similar to that described in Ref. [34]. The off-shell SM ggF Higgs boson 206 process, the $gg \rightarrow ZZ$ continuum, and their interference are considered as backgrounds. These samples 207 are generated at leading order (LO) in α_s using MCFM 6.1 [35] ($\ell\ell\ell\ell\ell$) or GG2vv 3.1.3 [36, 37] ($\ell\ell\nu\nu$) 208 but corrected to NNLO as a function of m_{ZZ} [38] using the same procedure as described in Ref. [6]. For 209 the $\ell \ell q q$ and $\nu \nu q q$ searches, the continuum $ZZ^{(*)}$ background is smaller so the $q\bar{q} \rightarrow ZZ^{(*)}$ sample is 210 used alone. It is scaled to include the contribution from $gg \rightarrow ZZ^{(*)}$ using the $gg \rightarrow ZZ^{(*)}$ cross-section 211 calculated by MCFM 6.1 [35]. 212

² The VBF channel is inclusive in quark flavour and hence dominated by the Z + light-quark jet background.

For samples in which the hard process is generated with ALPGEN or MC@NLO 4.03 [39], HERWIG 6.520 [40] is used to simulate parton showering and fragmentation, with JIMMY [41] used for the underlying-event simulation. PYTHIA 6.426 [42] is used for samples generated with MADGRAPH [43] and ACERMC [44], while PYTHIA 8.165 [45] is used for the GG2VV 3.1.3 [36, 37], MCFM 6.1 [46], and POWHEG samples. SHERPA implements its own parton showering and fragmentation model.

In the $\ell \ell q q$ and $\nu \nu q q$ searches, which have jets in the final state, the principal background is V + jets, 218 where V stands for either a W or a Z boson. In simulations of these backgrounds, jets are labelled 219 according to which generated hadrons with $p_{\rm T} > 5$ GeV are found within a cone of size $\Delta R = 0.4$ around 220 the reconstructed jet axis. If a *b*-hadron is found, the jet is labelled as a *b*-jet; if not and a charmed 221 hadron is found, the jet is labelled as a *c*-jet; if neither is found, the jet is labelled as a light (i.e., *u*-, *d*-, 222 or s-quark, or gluon) jet, denoted by 'j'. For V + jets events that pass the selections for these searches, 223 two of the additional jets are reconstructed as the hadronically-decaying Z boson candidate. Simulated 224 V + jets events are then categorized based on the labels of these jets. If one jet is labelled as a b-jet, the 225 event belongs to the V + b category; if not, and one of the jets is labelled as a c-jet, the event belongs to 226 the V + c category; otherwise, the event belongs to the V + i category. Further subdivisions are defined 227 according to the flavour of the other jet from the pair, using the same precedence order: V + bb, V + bc, 228 V + bj, V + cc, V + cj, and V + jj; the combination of V + bb, V + bc, and V + cc is denoted by V+hf. 229

3.4. Detector simulation

The simulation of the detector is performed with either a full ATLAS detector simulation [66] based on GEANT 4 9.6 [67] or a fast simulation³ based on a parameterization of the performance of the ATLAS electromagnetic and hadronic calorimeters [68] and on GEANT 4 elsewhere. All simulated samples are generated with a variable number of minimum-bias interactions (simulated using PYTHIA 8 with the MSTW2008LO PDF [69] and the A2 tune [48]), overlaid on the hard-scattering event to account for additional *pp* interactions in either the same or a neighbouring bunch crossing (pile-up).

²³⁷ Corrections are applied to the simulated samples to account for differences between data and simulation ²³⁸ for the lepton trigger and reconstruction efficiencies, and for the efficiency and misidentification rate of ²³⁹ the algorithm used to identify jets containing *b*-hadrons (*b*-tagging).

4. Object reconstruction and common event selection

The exact requirements used to identify physics objects vary between the different searches. This section outlines features that are common to all of the searches; search-specific requirements are given in the sections below.

Event vertices are formed from tracks with $p_{\rm T} > 400$ MeV. Each event must have an identified primary

vertex, which is chosen from among the vertices with at least three tracks as the one with the largest $\sum p_T^2$

²⁴⁶ of associated tracks.

³ The background samples that use the parameterized fast simulation are: SHERPA W/Z + jets production with $p_T^{W/Z} < 280$ GeV (for higher $p_T^{W/Z}$ the full simulation is used since it improves the description of the jet mass in the merged $\ell\ell qq$ search described in Section 7.1.2); POWHEG-BOX $t\bar{t}$, single top, and diboson production; and SM PYTHIA $q\bar{q} \rightarrow Zh$ and POWHEG-BOX $gg \rightarrow Zh$ production with $h \rightarrow bb$. The remaining background samples and the signal samples, with the exception of those used for the $\nu\nu qq$ search, use the full GEANT 4 simulation.

Physics process	$H \rightarrow ZZ$ search final state	Generator	Cross-section normalization	PDF set	Tune
	W/Z boson + jets				
$Z/\gamma^* \to \ell^+ \ell^- / \nu \bar{\nu}$	llll/llvv llqq [†] /vvqq llvv	Alpgen 2.14 [25] Sherpa 1.4.1 [24] Alpgen 2.14	NNLO [47] NNLO [49, 50] NNLO [47]	CTEQ6L1 [16] NLO CT10 CTEQ6L1	AUET2 [14, 48] Sherpa default AUET2
$W \to \ell \nu$	vvqq	Sherpa 1.4.1	NNLO [49, 50]	NLO CT10	Sherpa default
		Top quark			
tī	llll/llqq/vvqq llvv	Powheg-Box r2129 [51–53] MC@NLO 4.03 [39]	NNLO+NNLL [55, 56]	NLO CT10	Perugia2011C [54] AUET2
s-channel and Wt	llll/llqq/vvqq llvv	Powheg-Box r1556 MC@NLO 4.03	NNLO+NNLL [57, 58]	NLO CT10	Perugia2011C AUET2
<i>t</i> -channel	all	AcerMC 3.8 [44]	NNLO+NNLL [59]	CTEQ6L1	AUET2
		Dibosons			
$q\bar{q} \rightarrow ZZ(*)$	llqq/vvqq llll/llvv	Роwнед-Вох r1508 [60] Роwнед-Вох r1508 [60]	NLO [35, 61] NNLO QCD [31] NLO EW [32, 33]	NLO CT10 NLO CT10	AUET2 AUET2
$\begin{array}{l} \mathrm{EW}q\bar{q}\;(\rightarrow h)\rightarrow\\ ZZ(*)+2j \end{array}$	llll	MadGraph 5 1.3.28 [43]		CTEQ6L1	AUET2
$gg\;(\to h^*)\to ZZ$	ℓℓℓℓ ℓℓνν	MCFM 6.1 [46] GG2VV 3.1.3 [36, 37]	NNLO [38] (for $h \to ZZ$)	NLO CT10 NLO CT10	AU2 AU2
$q\bar{q} \rightarrow WZ$	llvv/llqq/vvqq llll	Powheg-Box r1508 Sherpa 1.4.1	NLO [35, 61]	NLO CT10	AUET2 Sherpa default
$q\bar{q} \rightarrow WW$	all	Powheg-Box r1508	NLO [35, 61]	NLO CT10	AUET2
		$m_h = 125 \text{ GeV SM Higgs boson}$	on (background) [‡]		
$\begin{array}{c} q\bar{q} \rightarrow Zh \rightarrow \\ \ell^{+}\ell^{-}b\bar{b}/v\bar{v}b\bar{b} \end{array}$	llqq/vvqq	Рутніа 8.165	NNLO [62-64]	CTEQ6L	AU2
$gg \to Zh \to \\ \ell^+ \ell^- b\bar{b} / \nu \bar{\nu} b\bar{b}$	$\ell \ell q q / \nu \nu q q$	Powheg-Box r1508	NLO [65]	CT10	AU2
		Signal			
$gg \to H \to ZZ(*)$ $q\bar{q} \to H + 2j;$ $H \to ZZ(*)$	all all	Powheg-Box r1508 Powheg-Box r1508		NLO CT10 NLO CT10	AU2 AU2
$q\bar{q} \rightarrow (W/Z)H;$ $H \rightarrow ZZ(*)$	lll	Рутніа 8.163	—	CTEQ6L1	AU2

Table 1: Details of the generation of simulated signal and background event samples. For each physics process, the table gives the final states generated, the $H \rightarrow ZZ$ final states(s) for which they are used, the generator, the PDF set, and the underlying-event tune. For the background samples, the order in α_S used to normalize the event yield is also given; for the signal, the normalization is the parameter of interest in the fit. More details can be found in the text. [†]The $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ VBF search uses ALPGEN instead.

[‡]For the $H \to ZZ \to \ell^+ \ell^- \ell^+ \ell^-$ and $H \to ZZ \to \ell^+ \ell^- \nu \bar{\nu}$ searches, the SM $h \to ZZ$ boson contribution, along with its interference with the continuum ZZ background, is included in the diboson samples.

Muon candidates ('muons') [70] generally consist of a track in the ID matched with one in the MS. 247 However, in the forward region (2.5 < $|\eta|$ < 2.7), MS tracks may be used with no matching ID tracks; 248 further, around $|\eta| = 0$, where there is a gap in MS coverage, ID tracks with no matching MS track 249 may be used if they match an energy deposit in the calorimeter consistent with a muon. In addition to 250 quality requirements, muon tracks are required to pass close to the reconstructed primary event vertex. 251 The longitudinal impact parameter, z_0 , is required to be less than 10 mm, while the transverse impact 252 parameter, d_0 , is required to be less than 1 mm to reject non-collision backgrounds. This requirement is 253 not applied in in the case of muons with no ID track. 254

Electron candidates ('electrons') [71–73] consist of an energy cluster in the EM calorimeter with $|\eta| < 2.47$ 255 matched to a track reconstructed in the inner detector. The energy of the electron is measured from the 256 energy of the calorimeter cluster, while the direction is taken from the matching track. Electron candidates 257 are selected using variables sensitive to the shape of the EM cluster, the quality of the track, and the 258 goodness of the match between the cluster and the track. Depending on the search, either a selection is 259 made on each variable sequentially or all the variables are combined into a likelihood discriminant. 260

Electron and muon energies are calibrated from measurements of $Z \rightarrow ee/\mu\mu$ decays [70, 72]. Electrons and muons must be isolated from other tracks, using $p_T^{\ell,isol}/p_T^{\ell} < 0.1$, where $p_T^{\ell,isol}$ is the scalar sum of the 261 262 transverse momenta of tracks within a $\Delta R = 0.2$ cone around the electron or muon (excluding the electron 263 or muon track itself), and $p_{\rm T}^{\ell}$ is the transverse momentum of the electron or muon candidate. The isolation 264 requirement is not applied in the case of muons with no ID track. For searches with electrons or muons in 265 the final state, the reconstructed lepton candidates must match the trigger lepton candidates that resulted 266

in the events being recorded by the online selection. 267

Jets are reconstructed [74] using the anti- k_t algorithm [75] with a radius parameter R = 0.4 operating on 268 massless calorimeter energy clusters constructed using a nearest-neighbour algorithm. Jet energies and 269 directions are calibrated using energy- and η -dependent correction factors derived using MC simulations. 270 with an additional calibration applied to data samples derived from in situ measurements [76]. A correction 271 is also made for effects of energy from pile-up. For jets with $p_{\rm T} < 50$ GeV within the acceptance of the ID 272 $(|\eta| < 2.4)$, the fraction of the summed scalar $p_{\rm T}$ of the tracks associated with the jet (within a $\Delta R = 0.4$ 273 cone around the jet axis) contributed by those tracks originating from the primary vertex must be at least 274 50%. This ratio is called the jet vertex fraction (JVF), and this requirement reduces the number of jet 275 candidates originating from pile-up vertices [77, 78]. 276

In the $\ell \ell q q$ search at large Higgs boson masses, the decay products of the boosted Z boson may be 277 reconstructed as a single anti- k_t jet with a radius of R = 0.4. Such configurations are identified using the 278 jet invariant mass, obtained by summing the momenta of the jet constituents. After the energy calibration, 279 the jet masses are calibrated, based on Monte Carlo simulations, as a function of jet p_T , η , and mass. 280

The missing transverse momentum, with magnitude $E_{\rm T}^{\rm miss}$, is the negative vectorial sum of the transverse 281 momenta of all clusters in the calorimeters with $|\eta| < 4.5$, calibrated appropriately based on their 282 identification as contributing to electrons, photons, hadronic decays of τ leptons, jets, or unassociated 283 calorimeter clusters, and all selected muons in the event [79]. Calorimeter deposits associated with muons 284 are subtracted from $E_{\rm T}^{\rm miss}$ to avoid double counting. 285

Jets containing *b*-hadrons (*b*-jets) can be discriminated from other jets ('tagged') based on the relatively 286

long lifetime of b-hadrons. Several methods are used to tag jets originating from the fragmentation of 287

a *b*-quark, including looking for tracks with a large impact parameter with respect to the primary event 288

- vertex, looking for a secondary decay vertex, and reconstructing a b-hadron $\rightarrow c$ hadron decay chain. 289
- For the $\ell\ell qq$ and $\nu\nu qq$ searches, this information is combined into a single neural-network discriminant 290

('MV1c'). This is a continuous variable that is larger for jets that are more like *b*-jets. A selection is then 291 applied that gives an efficiency of about 70%, on average, for identifying true b-jets, while the efficiencies 292 for accepting c-jets or light-quark jets are 1/5 and 1/140 respectively [30, 80–83]. The $\ell\ell\nu\nu$ search uses 293 an alternative version of this discriminant, 'MV1' [80], to reject background due to top-quark production; 294 compared with MV1c it has a smaller c-jet rejection. Tag efficiencies and mistag rates are calibrated 295 using data. For the purpose of forming the invariant mass of the *b*-jets, m_{bb} , the energies of tagged jets 296 are corrected to account for muons within the jets and an additional $p_{\rm T}$ -dependent correction is applied to 297 account for biases in the response due to resolution effects. 298

In channels which require two *b*-tagged jets in the final state, the efficiency for simulated events of the dominant Z + jets background to pass the tagging selection is low. To effectively increase the sizes of simulated samples, jets are 'truth tagged': each event is weighted by the flavour-dependent probability of the jets to actually pass the tagging selection.

5. $H \rightarrow ZZ \rightarrow \ell^+ \ell^- \ell^+ \ell^-$ event selection and background estimation

5.1. Event selection

The event selection and background estimation for the $H \to ZZ \to \ell^+ \ell^- \ell^+ \ell^- (\ell \ell \ell \ell)$ search is very similar to the analysis described in Ref. [84]. More details may be found there; a summary is given here.

Higgs boson candidates in the $\ell\ell\ell\ell\ell$ search must have two same-flavour, opposite-charge lepton pairs. 307 Muons must satisfy $p_T > 6$ GeV and $|\eta| < 2.7$, while electrons are identified using the likelihood 308 discriminant corresponding to the 'loose LH' selection from Ref. [73] and must satisfy $p_{\rm T} > 7$ GeV. 309 The impact parameter requirements that are made for muons are also applied to electrons, and electrons 310 (muons) must also satisfy a requirement on the transverse impact parameter significance, $|d_0|/\sigma_{d_0} < 6.5$ 311 (3.5). For this search, the track-based isolation requirement is relaxed to $p_T^{\ell,\text{isol}}/p_T^{\ell} < 0.15$ for both the 312 electrons and muons. In addition, lepton candidates must also be isolated in $E_{\rm T}^{\ell,\rm isol}$, the sum of the 313 transverse energies in calorimeter cells within a $\Delta R = 0.2$ cone around the candidate (excluding the 314 deposit from the candidate itself). The requirement is $E_{\rm T}^{\ell,\rm isol}/p_{\rm T}^{\ell} < 0.2$ for electrons, < 0.3 for muons with 315 a matching ID track, and < 0.15 for other muons. The three highest- $p_{\rm T}$ leptons in the event must satisfy, 316 in order, $p_T > 20$, 15, and 10 GeV. To ensure well-measured leptons, and reduce backgrounds containing 317 electrons from bremsstrahlung, same-flavour leptons must be separated from each other by $\Delta R > 0.1$, and 318 different-flavour leptons by $\Delta R > 0.2$. Jets that are $\Delta R < 0.2$ from electrons are removed. Final states in 319 this search are classified depending on the flavours of the leptons present: 4μ , $2e^{2}\mu$, $2\mu^{2}e$, and 4e. The 320 selection of lepton pairs is made separately for each of these flavour combinations; the pair with invariant 321 mass closest to the Z boson mass is called the leading pair and its invariant mass, m_{12} , must be in the 322 range 50–106 GeV. For the $2e^{2\mu}$ channel, the electrons form the leading pair, while for the $2\mu^{2e}$ channel 323 the muons are leading. The second, subleading, pair of each combination is the pair from the remaining 324 leptons with invariant mass m_{34} closest to that of the Z boson in the range $m_{\min} < m_{34} < 115$ GeV. 325 Here m_{\min} is 12 GeV for $m_{\ell\ell\ell\ell} < 140$ GeV, rises linearly to 50 GeV at $m_{\ell\ell\ell\ell} = 190$ GeV, and remains 326 at 50 GeV for $m_{\ell\ell\ell\ell} > 190$ GeV. Finally, if more than one flavour combination passes the selection, 327 which could happen for events with more than four leptons, the flavour combination with the highest 328 expected signal acceptance is kept; i.e., in the order: 4μ , $2e2\mu$, $2\mu 2e$, and 4e. For 4μ and 4e events, if an 329 opposite-charge same-flavour dilepton pair is found with $m_{\ell\ell}$ below 5 GeV, the event is vetoed in order to 330 reject backgrounds from J/ψ decays. 331

To improve the mass resolution, the four-momentum of any reconstructed photon consistent with having

been radiated from one of the leptons in the leading pair is added to the final state. Also, the four-momenta of the leptons in the leading pair are adjusted by means of a kinematic fit assuming a $Z \rightarrow \ell \ell$ decay; this improves the $m_{\ell\ell\ell\ell}$ resolution by up to 15%, depending on m_H . This is not applied to the subleading pair in order to retain sensitivity at lower m_H where one of the Z boson decays may be off-shell. For 4μ events, the resulting mass resolution varies from 1.5% at $m_H = 200$ GeV to 3.5% at $m_H = 1$ TeV, while for 4e events it ranges from 2% at $m_H = 200$ GeV to below 1% at 1 TeV.

Signal events can be produced via ggF or VBF, or associated production (VH, where V stands for either 339 a W or a Z boson). In order to measure the rates for these processes separately, events passing the event 340 selection described above are classified into channels, either ggF, VBF, or VH. Events containing at least 341 two jets with $p_T > 25$ GeV and $|\eta| < 2.5$ or $p_T > 30$ GeV and $2.5 < |\eta| < 4.5$ and with the leading two 342 such jets having $m_{ii} > 130$ GeV are classified as VBF events. Otherwise, if a jet pair satisfying the same 343 $p_{\rm T}$ and η requirements is present but with $40 < m_{ii} < 130$ GeV, the event is classified as VH, providing 344 it also passes a selection on a multivariate discriminant used to separate the VH and ggF signal. The 345 multivariate discriminant makes use of m_{ii} , $\Delta \eta_{ii}$, the $p_{\rm T}$ of the two jets, and the η of the leading jet. In 346 order to account for leptonic decays of the V (W or Z) boson, events failing this selection may still be 347 classified as VH if an additional lepton with $p_{\rm T} > 8$ GeV is present. All remaining events are classified as 348 ggF. Due to the differing background compositions and signal resolutions, events in the ggF channel are 349 further classified into subchannels according to their final state: 4e, $2e^{2\mu}$, $2\mu^{2e}$, or 4μ . The selection for 350 VBF is looser than is optimal, however the effect on the final results is fairly small (< 5% for large m_H). 351 The $m_{\ell\ell\ell\ell}$ distributions for the three channels are shown in Fig. 1. 352

5.2. Background estimation

The dominant background in this channel is continuum $ZZ^{(*)}$ production. Its contribution to the yield is 354 determined from simulation using the samples described in Section 3.3. Other background components 355 are small and consist mainly of $t\bar{t}$ and Z + jets events. These are difficult to estimate from MC simulations 356 due to the small rate at which such events pass the event selection, and also because they depend on 357 details of jet fragmentation, which are difficult to model reliably in simulations. Therefore, both the 358 rate and composition of these backgrounds are estimated from data. Since the composition of the these 359 backgrounds depends on the flavour of the subleading dilepton pair, different approaches are taken for the 360 $\ell\ell\mu\mu$ and the $\ell\ell ee$ final states. 361

The $\ell\ell\mu\mu$ non-ZZ background comprises mostly $t\bar{t}$ and $Z + b\bar{b}$ events, where in the latter the muons arise 362 mostly from heavy-flavour semileptonic decays, and to a lesser extent from π/K in-flight decays. The 363 contribution from single-top production is negligible. The normalization of each component is estimated 364 by a simultaneous fit to the m_{12} distribution in four control regions, defined by inverting the impact 365 parameter significance or isolation requirements on the subleading muon, or by selecting a subleading $e\mu$ 366 or same-charge pair. A small contribution from WZ decays is estimated using simulation. The electron 367 background contributing to the *llee* final states comes mainly from jets misidentified as electrons, arising 368 in three ways: light-flavour hadrons misidentified as electrons, photon conversions reconstructed as 369 electrons, and non-isolated electrons from heavy-flavour hadronic decays. This background is estimated 370 in a control region in which the three highest- $p_{\rm T}$ leptons must satisfy the full selection, with the third 371 lepton being an electron. For the lowest- $p_{\rm T}$ lepton, which must also be an electron, the impact parameter 372 and isolation requirements are removed and the likelihood requirement is relaxed. In addition, it must 373 have the same charge as the other subleading electron in order to minimize the contribution from the $ZZ^{(*)}$ 374



Figure 1: The distributions of $m_{\ell\ell\ell\ell}$ used in the likelihood fit for the $H \to ZZ \to \ell^+ \ell^- \ell^+ \ell^-$ search in the (a) ggF, (b) VBF, and (c) VH channels. The 'Z + jets, $t\bar{t}$ ' entry includes all backgrounds other than ZZ, as measured from data. No events are observed beyond the upper limit of the plots. The simulated $m_H = 200$ GeV signal is normalized to a cross-section corresponding to five times the observed limit given in Section 11 (a single limit is derived for the VBF and VH modes combined; the relative normalizations for these modes are taken from theoretical predictions). Figure (b) shows both the VBF and VH signal modes as there is significant contamination of VH events into the VBF category.

background. The yields of the background components of the lowest- $p_{\rm T}$ lepton are extracted with a fit

to the number of hits in the innermost pixel layer and the ratio of the number of high-threshold to low-

threshold TRT hits (which provides discrimination between electrons and pions). For both backgrounds,

the fitted yields in the control regions are extrapolated to the signal region using efficiencies obtained from

379 simulation.

For the non-ZZ components of the background, the $m_{\ell\ell\ell\ell}$ shape is evaluated for the $\ell\ell\mu\mu$ final states using simulated events, and from data for the $\ell\ell ee$ final states by extrapolating the shape from the $\ell\ell ee$ control region described above. The fraction of this background in each channel (ggF, VBF, *VH*) is evaluated using simulation. The non-ZZ background contribution for $m_{\ell\ell\ell\ell} > 140$ GeV is found to be approximately 4% of the total background.

Major sources of uncertainty in the estimate of the non-ZZ backgrounds include differences in the results 385 when alternative methods are used to estimate the background [84], uncertainties in the transfer factors 386 used to extrapolate from the control region to the signal region, and the limited statistical precision in 387 the control regions. For the $\ell\ell\mu\mu$ ($\ell\ell ee$) background, the uncertainty is 21% (27%) in the ggF channel. 388 100% (117%) in the VBF channel, and 62% (79%) in the VH channel. The larger uncertainty in the VBF 389 channel arises due to large statistical uncertainties on the fraction of Z + jets events falling in this channel. 390 Uncertainties in the expected $m_{\ell\ell\ell\ell}$ shape are estimated from differences in the shapes obtained using 391 different methods for estimating the background. 392

³⁹³ 6. $H \rightarrow ZZ \rightarrow \ell^+ \ell^- \nu \bar{\nu}$ event selection and background estimation

6.1. Event selection

The event selection for the $H \to ZZ \to \ell^+ \ell^- \nu \bar{\nu} (\ell \ell \nu \nu)$ search starts with the reconstruction of either a 395 $Z \to e^+e^-$ or $Z \to \mu^+\mu^-$ lepton pair; the leptons must be of opposite charge and must have invariant mass 396 $76 < m_{\ell\ell} < 106$ GeV. The charged lepton selection is tighter than that described in Section 4. Muons 397 must have matching tracks in the ID and MS and lie in the region $|\eta| < 2.5$. Electrons are identified using a 398 series of sequential requirements on the discriminating variables, corresponding to the 'medium' selection 399 from Ref. [73]. Candidate leptons for the $Z \to \ell^+ \ell^-$ decay must have $p_T > 20$ GeV, and leptons within a 400 cone of $\Delta R = 0.4$ around jets are removed. Jets that lie $\Delta R < 0.2$ of electrons are also removed. Events 401 containing a third lepton or muon with $p_{\rm T} > 7$ GeV are rejected; for the purpose of this requirement, the 402 'loose' electron selection from Ref. [73] is used. To select events with neutrinos in the final state, the 403 magnitude of the missing transverse momentum must satisfy $E_{\rm T}^{\rm miss} > 70$ GeV. 404

As in the $\ell\ell\ell\ell$ search, samples enriched in either ggF or VBF production are selected. An event is classified as VBF if it has at least two jets with $p_T > 30$ GeV and $|\eta| < 4.5$ with $m_{jj} > 550$ GeV and $\Delta \eta_{jj} > 4.4$. Events failing to satisfy the VBF criteria and having no more than one jet with $p_T > 30$ GeV and $|\eta| < 2.5$ are classified as ggF. Events not satisfying either set of criteria are rejected.

To suppress the Drell–Yan background, the azimuthal angle between the combined dilepton system and the missing transverse momentum vector $\Delta \phi(p_T^{\ell \ell}, E_T^{\text{miss}})$ must be greater than 2.8 (2.7) for the ggF (VBF) channel (optimized for signal significance in each channel), and the fractional p_T difference, defined as $|p_T^{\text{miss,jet}} - p_T^{\ell \ell}|/p_T^{\ell \ell}$, must be less than 20%, where $p_T^{\text{miss},jet} = |\vec{E}_T^{\text{miss}} + \sum_{j \in I} \vec{p}_T^{jet}|$. Z bosons originating from the decay of a high-mass state are boosted; thus, the azimuthal angle between the two leptons $\Delta \phi_{\ell \ell}$ must be less than 1.4. Events containing a *b*-tagged jet with $p_T > 20$ GeV and $|\eta| < 2.5$ are rejected in order to reduce the background from top-quark production. All jets in the event must have an azimuthal angle greater than 0.3 relative to the missing transverse momentum.

⁴¹⁷ The discriminating variable used is the transverse mass m_T^{ZZ} reconstructed from the momentum of the ⁴¹⁸ dilepton system and the missing transverse momentum, defined by:

$$(m_{\rm T}^{ZZ})^2 \equiv \left(\sqrt{m_Z^2 + \left|p_{\rm T}^{\ell\ell}\right|^2} + \sqrt{m_Z^2 + \left|E_{\rm T}^{\rm miss}\right|^2}\right)^2 - \left|\vec{p}_{\rm T}^{\ell\ell} + \vec{E}_{\rm T}^{\rm miss}\right|^2.$$
(1)

The resulting resolution in m_T^{ZZ} ranges from 7% at $m_H = 240$ GeV to 15% at $m_H = 1$ TeV.

Figure 2 shows the m_T^{ZZ} distribution in the ggF channel. The event yields in the VBF channel are very small (see Table 2).



Figure 2: The distribution of m_T^{ZZ} used in the likelihood fit for the $H \to ZZ \to \ell^+ \ell^- v \bar{v}$ search in the ggF channel. The simulated signal is normalized to a cross-section corresponding to five times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom pane shows the ratio of the observed data to the predicted background.

422 6.2. Background estimation

The dominant background is *ZZ* production, followed by *WZ* production. Other important backgrounds to this search include the *WW*, $t\bar{t}$, Wt, and $Z \rightarrow \tau^+\tau^-$ processes, and also the *Z* + jets process with poorly reconstructed $E_{\rm T}^{\rm miss}$, but these processes tend to yield final states with low $m_{\rm T}$. Backgrounds from *W* + jets, $t\bar{t}$, single top quark (*s*- and *t*-channel), and multijet processes with at least one jet misidentified as an electron or muon are very small.

The POWHEG simulation is used to estimate the *ZZ* background in the same way as for the $\ell\ell\ell\ell$ search. The *WZ* background is also estimated with POWHEG and validated with data using a sample of events that pass the signal selection and that contain an extra electron or muon in addition to the $Z \rightarrow \ell^+ \ell^-$ candidate.

The *WW*, $t\bar{t}$, *Wt*, and $Z \rightarrow \tau^+ \tau^-$ processes give rise to both same-flavour as well as different-flavour lepton final states. The total background from these processes in the same-flavour final state can be estimated from control samples that contain an electron–muon pair rather than a same-flavour lepton pair by

$$N_{ee}^{\text{bkg}} = \frac{1}{2} \times N_{e\mu}^{\text{data, sub}} \times \alpha,$$

$$N_{\mu\mu}^{\text{bkg}} = \frac{1}{2} \times N_{e\mu}^{\text{data, sub}} \times \frac{1}{\alpha},$$
(2)

where N_{ee}^{bkg} and $N_{\mu\mu}^{bkg}$ are the number of electron and muon pair events in the signal region and $N_{e\mu}^{data,sub}$ is the number of events in the $e\mu$ control sample with WZ, ZZ, and other small backgrounds (W + jets,

 $t\bar{t}W/Z$, and triboson) subtracted using simulation. The factor of two arises because the branching ratio 436 to final states containing electrons and muons is twice that of either ee or $\mu\mu$. The factor α takes into 437 account the different efficiencies for electrons and muons and is measured from data as $\alpha^2 = N_{ee}^{\text{data}}/N_{\mu\mu}^{\text{data}}$, 438 the ratio of the number of electron pair to muon pair events in the data after the Z boson mass requirement 439 $(76 < m_{\ell\ell} < 106 \text{ GeV})$. The measured value of α is 0.94 with a systematic uncertainty of 0.04 and a 440 negligible statistical uncertainty. There is also a systematic uncertainty from the background subtraction 441 in the control sample; this is found to be less than 1%. For the VBF channel, no events remain in the $e\mu$ 442 control sample after applying the full selection. In this case, the background estimate is calculated after 443 only the requirements on $E_{\rm T}^{\rm miss}$ and the number of jets; the efficiencies of the remaining selections for this 444 background are estimated using simulation. 445

The *Z* + jets background is estimated from data by comparing the signal region (A) with regions in which one (B, C) or both (D) of the $\Delta \phi_{\ell\ell}$ and $\Delta \phi(p_T^{\ell\ell}, E_T^{\text{miss}})$ requirements are reversed. An estimate of the number of background events in the signal region is then $N_A^{\text{est}} = N_C^{\text{obs}} \times (N_B^{\text{obs}}/N_D^{\text{obs}})$, where N_X^{obs} is the number of events observed in region *X* after subtracting non-*Z* boson backgrounds. The shape is estimated by taking N_C^{obs} (the region with the $\Delta \phi_{\ell\ell}$ requirement reversed) bin-by-bin and applying a correction derived from MC to account for shape difference between regions A and C. Systematic uncertainties arise from differences in the shape of the E_T^{miss} and m_T^{ZZ} distributions among the four regions, the small correlation between the two variables, and the subtraction of non-*Z* boson backgrounds.

The W + jets and multijet backgrounds are estimated from data using the fake-factor method [85]. This uses a control sample derived from data using a loosened requirement on $E_{\rm T}^{\rm miss}$ and several kinematic selections. The background in the signal region is then derived using an efficiency factor from simulation to correct for the acceptance. Both of these backgrounds are found to be negligible.

Table 2 shows the expected yields of the backgrounds and signal, and observed counts of data events. The
expected yields of the backgrounds in the table are after applying the combined likelihood fit to the data,
as explained in Section 10.

Process	ggF channel	VBF channel
$q\bar{q} \rightarrow ZZ$	$110 \pm 1 \pm 10$	$0.13 \pm 0.04 \pm 0.02$
$gg \rightarrow ZZ$	$11 \pm 0.1 \pm 5$	$0.12 \pm 0.01 \pm 0.05$
WZ	$47 \pm 1 \pm 5$	$0.10 \pm 0.05 \pm 0.1$
$WW/t\bar{t}/Wt/Z \rightarrow \tau^+ \tau^-$	$58 \pm 6 \pm 5$	$0.41 \pm 0.01 \pm 0.08$
$Z(\rightarrow e^+e^-, \mu^+\mu^-)$ +jets	$74 \pm 7 \pm 20$	$0.8 \pm 0.3 \pm 0.3$
Other backgrounds	$4.5 \pm 0.7 \pm 0.5$	i <u> </u>
Total background	$310 \pm 9 \pm 40$	$1.6 \pm 0.3 \pm 0.5$
Observed	309	4
ggF signal ($m_H = 400$ GeV)	$45 \pm 1 \pm 3$	
VBF signal ($m_H = 400 \text{ GeV}$)	$1 \pm < 0.1 \pm 2$	$10 \pm 0.5 \pm 1$

Table 2: Expected background yields and observed counts of data events after all selections for the ggF and VBF channels of the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- v \bar{v}$ search. The first and second uncertainties correspond to the statistical and systematic uncertainties, respectively.

⁴⁶¹ 7. $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q \bar{q}$ event selection and background estimation

462 7.1. Event selection

As in the previous search, the event selection starts with the reconstruction of a $Z \to \ell \ell$ decay. For 463 the purpose of this search, leptons are classified as either 'loose', with $p_{\rm T} > 7$ GeV, or 'tight', with 464 $p_{\rm T}$ > 25 GeV. Loose muons extend to $|\eta| < 2.7$, while tight muons are restricted to $|\eta| < 2.5$ and must 465 have tracks in both the ID and the MS. The transverse impact parameter requirement for muons is tightened 466 for this search to $|d_0| < 0.1$ mm. Electrons are identified using a likelihood discriminant very similar 467 to that used for the $\ell\ell\ell\ell\ell$ search, except that it was tuned for a higher signal efficiency. This selection 468 is denoted 'very loose LH' [73]. To avoid double counting, the following procedure is applied to loose 469 leptons and jets. First, any jets that lie $\Delta R < 0.4$ of an electron are removed. Next, if a jet is within 470 a cone of $\Delta R = 0.4$ of a muon, the jet is discarded if it has less than two matched tracks or if the JVF 471 recalculated without muons (see Section 4) is less than 0.5, since in this case it is likely to originate from a 472 muon having showered in the calorimeter; otherwise the muon is discarded. (Such muons are nevertheless 473 included in the computation of the $E_{\rm T}^{\rm miss}$ and in the jet energy corrections described in Section 4.) Finally, 474 if an electron is within a cone of $\Delta R = 0.2$ of a muon, the muon is kept unless it has no track in the MS. 475 in which case the electron is kept. 476

Events must contain a same-flavour lepton pair with invariant mass satisfying 83 $< m_{\ell\ell} < 99$ GeV. At least one of the leptons must be tight, while the other may be either tight or loose. Events containing any additional loose leptons are rejected. The two muons in a pair are required to have opposite charge, but this requirement is not imposed for electrons because larger energy losses from showering in material in the inner tracking detector lead to higher charge misidentification probabilities.

Jets used in this search to reconstruct the $Z \to q\bar{q}$ decay, referred to as 'signal' jets, must have $|\eta| < 2.5$ 482 and $p_{\rm T} > 20$ GeV; the leading signal jet must also have $p_{\rm T} > 45$ GeV. The search for forward jets in the 483 VBF production mode uses an alternative, 'loose', jet definition, which includes both signal jets and any 484 additional jets satisfying 2.5 < $|\eta|$ < 4.5 and p_T > 30 GeV. Since no high- p_T neutrinos are expected 485 in this search, the significance of the missing transverse momentum, $E_T^{\text{miss}}/\sqrt{H_T}$ (all quantities in GeV), 486 where $H_{\rm T}$ is the scalar sum of the transverse momenta of the leptons and loose jets, must be less than 487 3.5. This requirement is loosened to 6.0 for the case of the resolved channel (see Section 7.1.1) with 488 two b-tagged jets due to the presence of neutrinos from heavy-flavour decay. The $E_{\rm T}^{\rm miss}$ significance 489 requirement rejects mainly top-quark background. 490

Following the selection of the $Z \rightarrow \ell \ell$ decay, the search is divided into several channels: resolved ggF, merged-jet ggF, and VBF, as discussed below.

493 7.1.1. Resolved ggF channel

Over most of the mass range considered in this search ($m_H \leq 700$ GeV), the $Z \rightarrow q\bar{q}$ decay results in two well-separated jets that can be individually resolved. Events in this channel should thus contain at least two signal jets. Since *b*-jets occur much more often in the signal (~ 21% of the time) than in the dominant Z + jets background (~ 2% of the time), the sensitivity of this search is optimized by dividing it into 'tagged' and 'untagged' subchannels, containing events with exactly two and fewer than two *b*-tagged jets, respectively. Events with more than two *b*-tagged jets are rejected. In the tagged subchannel, the two tagged jets form the candidate $Z \rightarrow q\bar{q}$ decay. In the untagged subchannel, if there are no tagged jets, the two jets with largest transverse momenta are used. Otherwise, the *b*-tagged jet is paired with the untagged jet with the largest transverse momentum. The invariant mass of the chosen jet pair m_{jj} must be in the range 70–105 GeV in order to be consistent with $Z \rightarrow q\bar{q}$ decay. To maintain orthogonality, any events containing a VBF-jet pair as defined by the VBF channel (see Section 7.1.3) are excluded from the resolved selection.

The discriminating variable in this search is the invariant mass of the $\ell \ell j j$ system, $m_{\ell \ell j j}$; a signal should appear as a peak in this distribution. To improve the mass resolution, the energies of the jets forming the dijet pair are scaled event-by-event by a single multiplicative factor to set the dijet invariant mass m_{jj} to the mass of the Z boson (m_Z). This improves the resolution by a factor of 2.4 at $m_H = 200$ GeV. The resulting $m_{\ell \ell j j}$ resolution is 2–3%, approximately independent of m_H , for both the untagged and tagged channels.

Following the selection of the candidate $\ell \ell q q$ decay, further requirements are applied in order to optimize 512 the sensitivity of the search. For the untagged subchannel, the first requirement is on the transverse 513 momentum of the leading jet, p_T^j , which tends to be higher for the signal than for the background. The 514 optimal value for this requirement increases with increasing m_H . In order to avoid having distinct selections 515 for different m_H regions, p_T^j is normalized by the reconstructed final-state mass $m_{\ell\ell jj}$; the actual selection 516 is $p_T^j > 0.1 \times m_{\ell \ell j j}$. Studies have shown that the optimal requirement on $p_T^j / m_{\ell \ell j j}$ is nearly independent of 517 the assumed value of m_H . Second, the total transverse momentum of the dilepton pair also increases with 518 increasing m_H . Following a similar strategy, the selection is $p_T^{\ell\ell} > \min[-54 \text{ GeV} + 0.46 \times m_{\ell\ell i j}, 275 \text{ GeV}]$. 519 Finally, the azimuthal angle between the two leptons decreases with increasing m_H ; it must satisfy 520 $\Delta \phi_{\ell \ell} < (270 \text{ GeV}/m_{\ell \ell j j})^{3.5} + 1$. For the tagged channel, only one additional requirement is applied: 521 $p_{\rm T}^{\ell\ell} > \min[-79 \text{ GeV} + 0.44 \times m_{\ell\ell jj}, 275 \text{ GeV}];$ the different selection for $p_{\rm T}^{\ell\ell}$ increases the sensitivity of 522 the tagged channel at low m_H . Figures 3(a) and 3(b) show the $m_{\ell\ell jj}$ distributions of the two subchannels 523 after the final selection. 524

525 7.1.2. Merged-jet ggF channel

For very large Higgs boson masses, $m_H \gtrsim 700$ GeV, the Z bosons become highly boosted and the jets from $Z \rightarrow q\bar{q}$ decay start to overlap, causing the resolved channel to lose efficiency. The merged-jet channel recovers some of this loss by looking for a $Z \rightarrow q\bar{q}$ decay that is reconstructed as a single jet.

Events are considered for the merged-jet channel if they have exactly one signal jet, or if the selected jet pair has an invariant mass outside the range 50–150 GeV (encompassing both the signal region and the control regions used for studying the background). Thus, the merged-jet channel is explicitly orthogonal to the resolved channel.

To be considered for the merged-jet channel, the dilepton pair must have $p_T^{\ell \ell} > 280$ GeV. This not only 533 ensures that the Z bosons are highly boosted, but also ensures that the MC simulation for the Z + jets534 background consists mainly of the SHERPA samples with $p_T^Z > 280$ GeV, which were processed with the 535 full detector simulation (Section 3.4). The leading jet must also satisfy $p_{\rm T} > 200$ GeV and $m/p_{\rm T} > 0.05$, 536 where *m* is the jet mass, in order to restrict the jet to the kinematic range in which the mass calibration 537 has been studied. Finally, the invariant mass of the leading jet must be within the range 70–105 GeV. The 538 merged-jet channel is not split into subchannels based on the number of b-tagged jets; as the sample size 539 is small, this would not improve the expected significance. 540



Figure 3: The distributions of $m_{\ell\ell jj}$ used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ search in the (a) untagged and (b) tagged resolved ggF subchannels. The dashed line shows the total background used as input to the fit. The simulated signal is normalized to a cross-section corresponding to thirty times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.

Including this channel increases the overall efficiency for the $\ell \ell q q$ signal at $m_H = 900$ GeV by about a factor of two. Figure 4(a) shows the distribution of the invariant mass of the leading jet after all selections except for that on the jet invariant mass; it can be seen that the simulated signal has a peak at the mass of the Z boson, with a tail at lower masses due to events where the decay products of the Z boson are not fully contained in the jet cone. The discriminating variable for this channel is the invariant mass of the two leptons plus the leading jet, $m_{\ell\ell j}$, which has a resolution of 2.5% for a signal with $m_H = 900$ GeV and is shown in Fig. 4(b).

548 **7.1.3. VBF channel**

Events produced via the VBF process contain two forward jets in addition to the reconstructed leptons and signal jets from $ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ decay. These forward jets are called 'VBF jets'. The search in the VBF channel starts by identifying a candidate VBF-jet pair. Events must have at least four loose jets, two of them being non-*b*-tagged and pointing in opposite directions in *z* (that is, $\eta_1 \cdot \eta_2 < 0$). If more than one such pair is found, the one with the largest invariant mass, $m_{jj,\text{VBF}}$, is selected. The pair must further satisfy $m_{jj,\text{VBF}} > 500$ GeV and have a pseudorapidity gap of $|\Delta \eta_{jj,\text{VBF}}| > 4$. The distributions of these two variables are shown in Fig. 5.

Once a VBF-jet pair has been identified, the $ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ decay is reconstructed in exactly the same way as in the resolved channel, except that the jets used for the VBF-jet pair are excluded and no *b*-tagging categories are created due to the small sample size. The final $m_{\ell\ell jj}$ discriminant is shown in Fig. 6. Again,

DRAFT



Figure 4: Distributions for the merged-jet channel of the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ search after the mass calibration. (a) The invariant mass of the leading jet, m_j , after the kinematic selection for the $\ell\ell qq$ merged-jet channel. (b) The distribution of $m_{\ell\ell j}$ in the signal region used in the likelihood fit. It is obtained requiring $70 < m_j < 105$ GeV. The dashed line shows the total background used as input to the fit. The simulated signal is normalized to a cross-section corresponding to five times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background. The signal contribution is shown added on top of the background in (b) but not in (a).

the resolution is improved by constraining the dijet mass to m_Z as described in Section 7.1.1, resulting in a similar overall resolution of 2–3%.

561 7.2. Background estimation

The main background in the $\ell \ell q q$ search is Z + jets production, with significant contributions from both top-quark and diboson production in the resolved ggF channel, as well as a small contribution from multijet production in all channels. For the multijet background, the shape and normalization is taken purely from data, as described below. For the other background processes, the input is taken from simulation, with data-driven corrections for Z + jets and $t\bar{t}$ production. The normalizations of the Z + jets and top-quark backgrounds are left free to float and are determined in the final likelihood fit as described below and in Section 10.

The Z + jets MC sample is constrained using control regions that have the same selection as the signal 569 regions except that m_{ii} (m_i in the case of the merged-jet channel) lies in a region just outside of that selected 570 by the signal Z boson requirement. For the resolved channels, the requirement for the control region is 571 $50 < m_{ii} < 70$ GeV or $105 < m_{ii} < 150$ GeV; for the merged-jet channel, it is $30 < m_i < 70$ GeV. In the 572 resolved ggF channel, which is split into untagged and tagged subchannels as described in Section 7.1.1, 573 the Z + jets control region is further subdivided into 0-tag, 1-tag, and 2-tag subchannels based on the 574 number of b-tagged jets. The sum of the 0-tag and 1-tag subchannels is referred to as the untagged control 575 region, while the 2-tag subchannel is referred to as the tagged control region. 576



Figure 5: Distribution of (a) invariant mass and (b) pseudorapidity gap for the VBF-jet pair in the VBF channel of the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ search before applying the requirements on these variables (and prior to the combined fit described in Section 10). The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.



Figure 6: The distribution of $m_{\ell\ell jj}$ used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ search in the VBF channel. The dashed line shows the total background used as input to the fit. The simulated signal is normalized to a cross-section corresponding to thirty times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom pane shows the ratio of the observed data to the predicted background.

The normalization of the Z + jets background is determined by the final profile-likelihood fit as described 577 in Section 10. In the resolved ggF channel, the simulated Z + jets sample is split into several different 578 components according to the true flavour of the jets as described in Section 3.3: Z + jj, Z + cj, Z + bj, 579 and Z+hf. The individual normalizations for each of these four components are free to float in the fit 580 and are constrained by providing as input to the fit the distribution of the "b-tagging category" in the 581 untaged and tagged Z + jets control regions. The *b*-tagging category is formed from the combination of 582 the MV1c b-tagging discriminants of the two signal jets as described in Appendix B. In the VBF and 583 merged-jet ggF channels, which are not divided into b-tag subchannels, the background is dominated by 584 Z+light-jets. Thus, only the inclusive Z + jets normalization is varied in the fit for these channels. Since 585 these two channels probe very different regions of phase space, each has a separate normalization factor in 586 the fit; these are constrained by providing to the fit the distributions of $m_{\ell\ell i i}$ or $m_{\ell\ell i i}$ for the corresponding 587 Z + jets control regions. 588

⁵⁸⁹ Differences are observed between data and MC simulation for the distributions of the azimuthal angle ⁵⁸⁰ between the two signal jets, $\Delta \phi_{jj}$, and the transverse momentum of the leptonically-decaying Z boson, $p_T^{\ell\ell}$, ⁵⁹¹ for the resolved region, and for the $m_{\ell\ell jj}$ distribution in the VBF channel. To correct for these differences, ⁵⁹² corrections are applied to the SHERPA Z + jets simulation (prior to the likelihood fit) as described in ⁵⁹³ Appendix A. The distributions of $m_{\ell\ell jj}$ or $m_{\ell\ell j}$ in the various Z + jets control regions are shown in Fig. 7; ⁵⁹⁴ it can be seen that after the corrections (and after normalizing to the results of the likelihood fit), the ⁵⁹⁵ simulation provides a good description of the data.

The simulation models the m_{jj} distribution well in the resolved ggF and VBF channels. An uncertainty is assigned by weighting each event of the Z + jets MC simulation by a linear function of m_{jj} in order to cover the residual difference between data and MC events in the control regions.

Top-quark production is a significant background in the tagged subchannel of the resolved ggF channel. 599 This background is predominantly (> 97%) $t\bar{t}$ production with only a small contribution from single-top 600 processes, mainly Wt production. Corrections to the simulation to account for discrepancies in the p_T^{T} 601 distributions are described in Appendix A. The description of the top-quark background is cross-checked 602 and normalized using a control region with a selection identical to that of the tagged ggF channel except 603 that instead of two same-flavour leptons, events must contain an electron and a muon with opposite charge. 604 The $m_{\ell\ell j j}$ distribution in this control region is used as an input to the final profile-likelihood fit, in which 605 the normalization of the top-quark background is left free to float (see Section 10). There are few events 606 in the control region for the VBF and merged-jet ggF channels, so the normalization is assumed to be the 607 same across all channels, in which the top-quark contribution to the background is very small. Figure 8 608 shows that the data in the control region are well-described by the simulation after the normalization. 609

Further uncertainties in the top-quark background arising from the parton showering and hadronization models are estimated by varying the amount of parton showering in ACERMC and also by comparing with POWHEG+HERWIG. Uncertainties in the $t\bar{t}$ production matrix element are estimated by comparing the leading-order MC generator ALPGEN with the NLO generator aMC@NLO. Comparisons are also made with alternate PDF sets. A similar procedure is used for single-top production. In addition, for the dominant Wt single-top channel, uncertainties in the shapes of the m_{jj} and leading-jet p_T distributions are evaluated by comparing results from HERWIG to those from ACERMC.

The small multijet background in the $H \rightarrow ZZ \rightarrow eeqq$ decay mode is estimated from data by selecting a sample of events with the electron isolation requirement inverted, which is then normalized by fitting the m_{ee} distribution in each channel. In the $H \rightarrow ZZ \rightarrow \mu \mu qq$ decay mode, the multijet background is found to be perfigible. The residual multijet background in the ten querk control region is taken from the

found to be negligible. The residual multijet background in the top-quark control region is taken from the



Figure 7: The distributions of $m_{\ell\ell jj}$ or $m_{\ell\ell j}$ in the Z + jets control region of the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ search in the (a) untagged ggF, (b) tagged ggF, (c) merged-jet ggF, and (d) VBF channels. The dashed line shows the total background used as input to the fit. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.

opposite-charge $e\mu$ data events, which also accounts for the small W + jets background in that region. An uncertainty of 50% is assigned to these two normalizations, which are taken to be uncorrelated.

⁶²³ The diboson background, composed mainly of ZZ and $WZ \rightarrow \ell \ell j j$ production, and the SM $Zh \rightarrow \ell \ell bb$

background are taken directly from Monte Carlo simulation, as described in Section 3.3. The uncertainty in

the diboson background is estimated by varying the factorization and renormalization scales in an MCFM



DRAFT

Figure 8: The distribution of $m_{\ell\ell jj}$ in the $e\mu$ top-quark control region of the $H \to ZZ \to \ell^+ \ell^- q\bar{q}$ search in the tagged ggF channel. The dashed line shows the total background used as input to the fit. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom pane shows the ratio of the observed data to the predicted background.

calculation [35]. The method described in Refs. [86, 87] is used to avoid underestimating the uncertainty due to cancellations. Differences due to the choice of alternate PDF sets and variations in the value of $\alpha_{\rm S}$ are included in the normalization uncertainty. Additional shape uncertainties in the m_{ii} distribution

are obtained by comparing results from HERWIG, an LO simulation, with those from PowHEG+PYTHIA, an

630 NLO simulation.

The rate of the SM $Vh(V = W/Z, h \rightarrow bb)$ process, relative to the SM expectation, has been measured by ATLAS as $\mu = \sigma/\sigma_{SM} = 0.52 \pm 0.32$ (stat.) ± 0.24 (syst.) [30]. Since this is compatible with the SM expectation, the small $Zh(h \rightarrow bb)$ background in this channel is normalized to the SM cross-section and a 50% uncertainty is assigned to cover the difference between the prediction and the measured mean value.

8. $H \rightarrow ZZ \rightarrow v\bar{v}q\bar{q}$ event selection and background estimation

637 8.1. Event selection

Events selected for this search must contain no electrons or muons as defined by the 'loose' lepton selection of the $\ell\ell qq$ search. To select events with neutrinos in the final state, the magnitude of the missing transverse momentum vector must satisfy $E_T^{miss} > 160$ GeV; the trigger is 100% efficient in this range. Events must have at least two jets with $p_T > 20$ GeV and $|\eta| < 2.5$; the leading jet must further satisfy $p_T > 45$ GeV. To select a candidate $Z \rightarrow q\bar{q}$ decay, the invariant mass of the leading two jets must satisfy $70 < m_{jj} < 105$ GeV.

The multijet background, due mainly to the mismeasurement of jet energies, is suppressed using a track-based missing transverse momentum, \vec{p}_{T}^{miss} , defined as the negative vectorial sum of the transverse momenta of all good-quality inner detector tracks. The requirements are $p_T^{\text{miss}} > 30$ GeV, the azimuthal angle between the directions of \vec{E}_T^{miss} and \vec{p}_T^{miss} satisfy $\Delta \phi(\vec{E}_T^{\text{miss}}, \vec{p}_T^{\text{miss}}) < \pi/2$, and the azimuthal angle between the directions of \vec{E}_T^{miss} and the nearest jet satisfy $\Delta \phi(\vec{E}_T^{\text{miss}}, j) > 0.6$.

As in the resolved ggF channel of the $\ell\ell qq$ search, this search is divided into 'tagged' (exactly two *b*-tagged jets) and 'untagged' (fewer than two *b*-tagged jets) subchannels. Events with more than two *b*-tags are rejected.

The sensitivity of this search is improved by adding a requirement on the jet transverse momenta. As in 652 the $\ell \ell q q$ search, the optimal threshold depends on m_H . However, due to the neutrinos in the final state, 653 this decay mode does not provide a good event-by-event measurement of the mass of the diboson system, 654 m_{ZZ} . So, rather than having a single requirement on the jet transverse energy which is a function of the 655 measured m_{ZZ} , instead there is a set of requirements, based on the generated m_H , with the background 656 estimated separately for each of these separate jet requirements. The specific requirement is found by 657 rounding the generated m_H to the nearest 100 GeV; this is called m_H^{bin} . Then the subleading jet must 658 satisfy $p_T^{j2} > 0.1 \times m_H^{\text{bin}}$ in events with no *b*-tagged jets, and $p_T^{j2} > 0.1 \times m_H^{\text{bin}} - 10$ GeV in events with at 659 least one *b*-tagged jet. 660

The discriminating variable for this search is the transverse mass of the $\nu\nu qq$ system, shown in Fig. 9, defined as in Eq. (1) with p_T^{jj} replacing $p_T^{\ell\ell}$. To improve the transverse mass resolution, the energies of the leading two jets are scaled event-by-event by a multiplicative factor to set the dijet invariant mass m_{jj} to the Z boson mass, in the same manner as in the $\ell\ell qq$ search. This improves the transverse mass resolution by approximately 20% at $m_H = 400$ GeV and by approximately 10% at $m_H = 1$ TeV. The resulting resolution in m_T ranges from about 9% at $m_H = 400$ GeV to 14% at $m_H = 1$ TeV.

8.2. Background estimation

The dominant backgrounds for this search are Z + jets, W + jets, and $t\bar{t}$. The Z + jets background is estimated using a control sample selected in the same way as the signal sample except that events must contain exactly two loose muons. The $E_{\rm T}^{\rm miss}$ is calculated without including the muons and must satisfy the same requirement as for the signal: $E_{\rm T}^{\rm miss no \ \mu} > 160$ GeV. The Z + jets MC simulation is corrected as a function of $\Delta \phi_{jj}$ and $p_{\rm T}^{\ell\ell}$ in the same manner as in the resolved ggF channel of the $\ell\ell qq$ search, as described in Section 7.2 and Appendix A.

The W + jets background estimate similarly uses a control sample with the same selection as the signal sample except that there must be exactly one loose muon and the $E_{\rm T}^{\rm miss}$ requirement is again on $E_{\rm T}^{\rm miss}$. As in the case of Z + jets background, the flavour components of the W + jets background are free to float in the fit, and are constrained by providing as input to the fit the distribution of the *b*-tagging category,

described in Appendix B, in the 0-*b*-tag and 1-*b*-tag control regions.

Agreement between simulation and data for this background is improved by applying the correction on $\Delta \phi_{jj}$ for W + j and W + cj, with half the correction assigned as a systematic uncertainty, in the same manner as in Ref. [30]. For W+heavy-flavour-jets, no correction is applied, but a dedicated systematic

⁶⁸² uncertainty is assigned as described in Ref. [30].

Even after these corrections, the simulation does not accurately describe the data in the Z + jets and W + jets control sample with no *b*-tagged jets (which is dominated by Z/W + jj) for important kinematic

distributions such as $E_{\rm T}^{\rm miss}$ and jet transverse momenta. Moreover, because the resolution of the transverse



Figure 9: The distributions of m_T , the transverse mass of the $Z(\nu\nu)Z(jj)$ system, used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \nu \bar{\nu} q \bar{q}$ search in the (a, c) untagged and (b, d) tagged channels, for Higgs boson mass hypotheses of (a, b) $m_H = 400$ GeV and (c, d) $m_H = 900$ GeV. The dashed line shows the total background used as input to the fit. For the $m_H = 400$ GeV hypothesis (a, b) the simulated signal is normalized to a cross-section corresponding to twenty times the observed limit given in Section 11, while for the $m_H = 900$ GeV hypothesis (c, d) it is normalized to thirty times the observed limit. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.

mass of the $ZZ \rightarrow v\bar{v}q\bar{q}$ system is worse than that of $m_{\ell\ell jj}$, the vvqq search is more sensitive to $E_{\rm T}^{\rm miss}$ (i.e.

- $_{687}$ Z/W boson $p_{\rm T}$) than the $\ell\ell qq$ search. Therefore, a further correction is applied, as a linear function of $E_{\rm T}^{\rm miss}$, derived from measuring the ratio of the $E_{\rm T}^{\rm miss}$ distributions from simulation and data in the control
 - 30th June 2015 15:00

sample with no *b*-tagged jets after non-Z/W + jj backgrounds have been subtracted. An uncertainty of 50% is assigned to this correction. Following this correction, there is good agreement between simulation and data, as shown in Figs. 10 and 11.



Figure 10: The distributions of (a) $E_{\rm T}^{\rm miss}$ and (b) leading-jet $p_{\rm T}$ from the untagged $(Z \rightarrow \mu \mu)$ + jets control sample of the $H \rightarrow ZZ \rightarrow v \bar{v} q \bar{q}$ search. The dashed line shows the total background used as input to the fit. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.

The $t\bar{t}$ background is treated in the same manner as in the $\ell\ell qq$ search; in particular, $p_T^{t\bar{t}}$ is corrected in the same way.

Backgrounds from diboson and single-top production are estimated directly from MC simulations, both for shapes and normalization. The multijet background is estimated using a method similar to that used for the Z + jets background in the $\ell\ell\nu\nu$ search (Section 6.2), except that the variables used are $\Delta\phi(\vec{E}_{T}^{miss}, \vec{p}_{T}^{miss})$ and $\Delta\phi(\vec{E}_{T}^{miss}, j)$ [30]. It is found to be negligible.

9. Systematic uncertainties

The systematic uncertainties can be divided into three categories: experimental uncertainties, related to the detector or to the reconstruction algorithms, uncertainties in the modelling of the signal, and uncertainties in the estimation of the backgrounds. The first two are largely common to all the searches and are treated as fully correlated. The uncertainties in the estimates of most backgrounds vary from search to search, and are summarized in the background estimation sections above. The estimation of the uncertainty of the $ZZ^{(*)}$ background is outlined in Section 9.3.



Figure 11: The distributions of (a) $E_{\rm T}^{\rm miss}$ and (b) leading-jet $p_{\rm T}$ from the untagged $(W \to \mu \nu)$ + jets control sample of the $H \to ZZ \to \nu \bar{\nu} q \bar{q}$ search. The dashed line shows the total background used as input to the fit. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.

705 9.1. Experimental uncertainties

The following detector-related systematic uncertainties are common to all the searches unless otherwise stated.

The uncertainty in the integrated luminosity is determined to be 2.8% in a calibration following the methodology detailed in Ref. [88] using beam-separation scans performed in November 2012. This uncertainty is applied to the normalization of the signal and also to backgrounds for which the normalization is derived from MC calculations, and is correlated between all of the searches. There is also an uncertainty of 4% in the average number of interactions per bunch crossing, which leads to an uncertainty on distributions sensitive to pile-up.

There are small systematic uncertainties of O(1%) in the reconstruction and identification efficiencies for electrons and muons [70–73]. For the vvqq search, the uncertainty is instead in the efficiency of the lepton veto, and is also O(1%). Uncertainties in the lepton energy scale and resolution are also taken into account. These uncertainties are treated as uncorrelated between all of the searches due to differences in lepton selections optimized for each search.

The uncertainty in the jet energy scale has several sources, including uncertainties in the in situ calibration analysis, corrections for pile-up, and the flavour composition of the sample [76, 89]. These uncertainties are decomposed into independent components. For central jets, the total relative uncertainty on the jet energy scale ranges from about 3% for jets with a $p_{\rm T}$ of 20 GeV to about 1% for a $p_{\rm T}$ of 1 TeV. The calibration of the *b*-jet transverse energy has an additional uncertainty of 1–2%. There is also an uncertainty in the jet energy resolution [90], which ranges from 10–20% for jets with a $p_{\rm T}$ of 20 GeV to less than 5% for jets with $p_{\rm T} > 200$ GeV. The uncertainty associated with the pile-up rejection requirement (Section 4) is evaluated by varying the nominal value of 50% between 47% and 53% [78]. The jet energy scale uncertainties are correlated between the $\ell \ell q q$ and $\nu \nu q q$ searches, and separately between the $\ell \ell \ell \ell$ and $\ell \ell \nu \nu$ searches. They are not correlated between the two pairs of searches because although the $\ell \ell q q$ and $\nu \nu q q$ control regions have the power to constrain the jet energy scale uncertainties, these constraints do not necessarily apply to the $\ell \ell \ell \ell$ and $\ell \ell \nu \nu$ searches due to differences in the jet kinematics and composition.

⁷³² Uncertainties on the lepton and jet energy scales are propagated into the uncertainty on E_T^{miss} . A ⁷³³ contribution to E_T^{miss} also comes from energy deposits that are not associated with any identified physics ⁷³⁴ object; uncertainties on the energy calibration (8%) and resolution (3%) of the sum of these deposits are ⁷³⁵ also propagated to the uncertainty on E_T^{miss} [91].

⁷³⁶ Uncertainties in the efficiency for tagging *b*-jets and in the rejection factor for light jets are determined ⁷³⁷ from $t\bar{t}$ and dijet control samples [81–83]. Additional uncertainties account for differences in *b*-tagging ⁷³⁸ efficiency between simulated samples generated with SHERPA and PYTHIA and for differences observed ⁷³⁹ between standard *b*-tagging and truth tagging (defined at the end of Section 4) for close-by jets [30].

The efficiencies for the lepton triggers in events with reconstructed leptons are nearly 100%, and hence the related uncertainties are negligible. For the selection used in the $\nu\nu qq$ search, the efficiency for the Trigger is also close to 100% with negligible associated uncertainties.

The merged-jet channel of the $\ell\ell qq$ search relies on measuring single-jet masses. To estimate the 743 uncertainty in this measurement, jets reconstructed as described in Section 4 are compared with jets 744 constructed using the same clustering algorithm but using as input charged-particle tracks rather than 745 calorimeter energy deposits. The uncertainty is found using a procedure similar to that described in 746 Ref. [92] by studying the double ratio of masses of jets found by both the calorimeter- and track-based algorithms: $R_{\text{trackcalo}}^m = r_{\text{trackcalo}}^{m,\text{MC}}/r_{\text{trackcalo}}^{m,\text{MC}}$, where $r_{\text{trackcalo}}^m = m_{\text{calo}}^X/m_{\text{track}}^X$, X = data or MC simulation, and m is the jet mass. The uncertainty is taken as the deviation of this quantity from unity. Studies performed 747 748 749 on dijet samples yield a constant value of 10% for this uncertainty. Applying the jet mass calibration 750 derived from single jets in generic multijet samples to merged jets originating from boosted Z bosons 751 results in a residual topology-dependent miscalibration. This effect can be bounded by an additional 752 uncertainty of 10%. Adding these two effects in quadrature gives a total uncertainty on the jet mass scale 753 of 14%. The uncertainty on the jet mass resolution has a negligible effect on the final result. 754

9.2. Signal acceptance uncertainty

The uncertainty in the experimental acceptance for the Higgs boson signal due to the modelling of Higgs boson production is estimated by varying parameters in the generator and re-applying the signal selection at generator level. The renormalization and factorization scales are varied up and down both independently and coherently by a factor of two; the amounts of initial- and final-state radiation (ISR/FSR) are increased and decreased separately; and the PDF set used is changed from the nominal CT10 to either MSTW2008 or NNPDF23.

9.3. ZZ^(*) background uncertainties

⁷⁶³ Uncertainties on the $ZZ^{(*)}$ background are treated as correlated between the $\ell\ell\ell\ell\ell$ and $\ell\ell\nu\nu$ searches.

Uncertainties in the PDF and in α_S are taken from Ref. [93] and are derived separately for the $q\bar{q} \rightarrow ZZ$ 764 and $gg \rightarrow ZZ$ backgrounds, using the envelope of the CT10, MSTW, and NNPDF error sets following 765 the PDF4LHC prescription given in Refs. [94, 95], giving an uncertainty parameterized in m_{ZZ} . These 766 uncertainties amount to 3% for the $q\bar{q} \rightarrow ZZ$ process and 8% for the $gg \rightarrow ZZ$ process and are found 767 to be anti-correlated between the two processes; this is taken into account in the fit. The QCD scale 768 uncertainty for the $q\bar{q} \rightarrow ZZ$ process is also taken from Ref. [93] and is based on varying the factorization 769 and renormalization scales up and down by a factor of two, giving an uncertainty parameterized in m_{ZZ} 770 amounting to 4% on average. 771

The deviation of the NLO electroweak *K*-factor from unity is varied up and down by 100% in events with high QCD activity or with an off-shell *Z* boson, as described in Ref. [96]; this leads to an additional overall uncertainty of 1–3% for the $q\bar{q} \rightarrow ZZ$ process.

Full NLO and NNLO QCD calculations exist for the $gg \rightarrow h^* \rightarrow ZZ^{(*)}$ process, but not for the $gg \rightarrow ZZ$ 775 continuum process. However, Ref. [97] showed that higher-order corrections affect $gg \rightarrow WW$ and 776 $gg \to h^* \to WW$ similarly, within a 30% uncertainty on the interference term. This yields about a 60% 777 uncertainty on the $gg \rightarrow WW$ process. Furthermore, Ref. [97] states that this conclusion also applies to 778 the $ZZ^{(*)}$ final state, so the gg-induced part of the off-shell light Higgs boson K-factor from Ref. [38] 779 is applied to the $gg \rightarrow ZZ$ background. The uncertainty on this K-factor depends on m_{ZZ} and is about 780 30%. An additional uncertainty of 100% is assigned to this procedure; this covers the 60% mentioned 781 above. This uncertainty corresponds to the range considered for the $gg \rightarrow ZZ$ background K-factor in 782 the ATLAS off-shell Higgs boson signal-strength measurement described in Ref. [96]. 783

Acceptance uncertainties for the ggF and VBF (and VH for $\ell\ell\ell\ell\ell$) channels due to the uncertainty on the 784 \leq 1-jet and 2-jet cross-sections are estimated for the $q\bar{q} \rightarrow ZZ$ background by comparing the acceptance 785 upon varying the factorization and renormalization scales and changing the PDF set. For $\ell\ell\ell\ell\ell$ this leads to 786 uncertainties of 4%, 8%, and 3% on the ggF, VBF, and VH channels, respectively, where the uncertainty 787 is fully anti-correlated between the ggF channel and the VBF and VH channels. For the $gg \rightarrow ZZ$ process 788 where only LO generators are available, the VBF jets are simulated only in the parton shower, and so 789 the acceptance uncertainty is estimated by taking the difference between the acceptances predicted by 790 MCFM+PYTHIA8 and SHERPA, which have different parton shower simulations; this amounts to 90% for 791 the VH channel. 792

10. Combination and statistical interpretation

The statistical treatment of the data is similar to that described in Refs. [98-102], and uses a simultaneous 794 profile-likelihood-ratio fit to the data from all of the searches. The parameter of interest is the cross-795 section times branching ratio for heavy Higgs boson production, assumed to be correlated between all 796 of the searches. It is assumed that an additional Higgs boson would be produced predominantly via the 797 ggF and VBF processes but that the ratio of the two production mechanisms is unknown in the absence 798 of a specific model. For this reason, fits for the ggF and VBF production processes are done separately. 799 and in each case the other process is allowed to float in the fit as an additional nuisance parameter. The 800 VH production mechanism is included in the fit for the $\ell\ell\ell\ell$ search and is assumed to scale with the VBF 801 signal since both the VH and VBF production mechanisms depend on the coupling of the Higgs boson to 802 vector bosons. 803

Search	C	hannel	SR	$Z \operatorname{CR}$	W CR	Top CR
lll	ggF VBF		т _{ееее} , т _{µµµµ} , т _{ееµµ} , т _{µµее} т _{{{{}}{{}}{{}}{{}}{{}}{{}}{{}}{{}}{{}}{}			
	VH		$m_{\ell\ell\ell\ell}$			
llvv	ggF		$m_{\mathrm{T}}^{ee}, m_{\mathrm{T}}^{\mu\mu}$			
	VBF		$N_{\rm evt}^{ee}, N_{\rm evt}^{\mu\mu}$			
llaa	ggF	untagged tagged	m _{ℓℓjj} m _{ℓℓjj}	MV1c MV1c		$m_{\ell\ell jj}$
iiqq	VBF	merged-jet	$m_{\ell\ell j} \ m_{\ell\ell j j}$	m _{ℓℓj} m _{ℓℓjj}		
vvqq	ggF	untagged	<i>m</i> _T		MV1c (0 <i>b</i> -tags) MV1c (1 <i>b</i> -tag)	
		tagged	m _T			

DRAFT

Table 3: Summary of the distributions entering the likelihood fit for each channel of each search, both in the signal region (SR) and the various control regions (CR) used to constrain the background. Each entry represents one distribution; some channels have several distributions for different lepton flavours. The distributions are unbinned for the $\ell\ell\ell\ell\ell$ search and binned elsewhere. The VBF channels of the $\ell\ell\nu\nu$ search use only the overall event counts. See the text for the definitions of the specific variables used as well as for the definitions of the signal and control regions.

The simultaneous fit proceeds as follows. For each channel of each search, there is a distribution of 804 the data with respect to some discriminating variable; these distributions are fitted to a sum of signal 805 and backgrounds. The particular variables used are summarized in Table 3. The distributions for the 806 $\ell\ell\ell\ell\ell$ search are unbinned, since the resolution of $m_{\ell\ell\ell\ell}$ is very good, while other searches have binned 807 distributions. For the VBF channels of the $\ell\ell\nu\nu$ search, only the overall event counts are used, rather 808 than distributions, as the sample sizes are very small. The $\ell \ell q q$ and $\nu \nu q q$ searches include additional 809 distributions in control regions in order to constrain the background, using either distributions of the 810 mass variable or of the MV1c b-tagging discriminant. The details of the specific variables used and the 811 definitions of the signal and control regions are discussed in Sections 5 to 8. 812

As discussed in Section 9, the signal acceptance uncertainties, and many of the background theoretical 813 and experimental uncertainties, are treated as fully correlated between the searches. A given correlated 814 uncertainty is modelled in the fit by using a nuisance parameter common to all of the searches. The 815 mass hypothesis for the heavy Higgs boson strongly affects which sources of systematic uncertainty 816 have the greatest effect on the result. At lower masses, the $ZZ^{(*)}$ background theory uncertainties, the 817 Z + jets modelling uncertainties, and the uncertainties on the jet energy scale dominate. At higher masses, 818 uncertainties in the $\ell\ell\nu\nu$ non-ZZ background, the jet mass scale, and the Z + jets background in the 819 merged-jet regime dominate. The contribution to the uncertainty on the best-fit signal cross-section from 820 the dominant systematic uncertainties is shown in Table 4. 821

DRAFT	
-------	--

ggF mode		VBF mode	
Systematic source E	ffect [%]	Systematic source E	ffect [%]
	$m_H =$	200 GeV	
$gg \rightarrow ZZ K$ -factor uncertainty	27	$gg \rightarrow ZZ$ acceptance	13
Z +hf $\Delta \phi$ reweighting	5.3	Jet vertex fraction $(\ell \ell q q / \nu v q q)$	13
Luminosity	5.2	$gg \rightarrow ZZ K$ -factor uncertainty	13
Jet energy resolution $(\ell \ell q q / \nu v q q)$	3.9	Z + jets $\Delta \phi$ reweighting	7.9
QCD scale $gg \rightarrow ZZ$	3.7	Jet energy scale η modelling ($\ell \ell q q / \nu \nu q q$	y) 5.3
$m_H = 400 \text{ GeV}$			
$qq \rightarrow ZZ PDF$	21	Z + jets estimate ($\ell\ell\nu\nu$)	34
QCD scale $qq \rightarrow ZZ$	13	Jet energy resolution $(\ell \ell \ell \ell \ell \ell \nu \nu)$	6.5
Z + jets estimate ($\ell\ell\nu\nu$)	13	VBF Z + jets $m_{\ell\ell j j}$	5.5
Signal acceptance ISR/FSR (<i>lllll</i> v	v) 7.8	Jet flavour composition $(\ell \ell \ell \ell \ell \ell \nu \nu)$	5.3
$Z + b\bar{b}, Z + c\bar{c}, p_{\mathrm{T}}^{\ell\ell}$	5.6	Jet vertex fraction $(\ell \ell q q / \nu \nu q q)$	4.8
	$m_H =$	900 GeV	
Jet mass scale $(\ell \ell q q)$	7	Z + jets estimate ($\ell\ell\nu\nu$)	19
$Z + jj p_{\rm T}^{\ell\ell}$ shape	5.6	Jet mass scale $(\ell \ell q q)$	8.7
$qq \rightarrow ZZ$ PDF	4.3	$Z + jj p_{\rm T}^{\ell\ell}$ shape	7.3
QCD scale $qq \rightarrow ZZ$	3.5	Jet energy resolution $(\ell \ell \ell \ell \ell \ell \nu \nu)$	4.4
Luminosity	2.6	Jet flavour composition (VV/Signal)	2.6

Table 4: The effect of the leading systematic uncertainties on the best-fit signal cross-section uncertainty, expressed as a percentage of the total (systematic and statistical) uncertainty, for the ggF (left) and VBF (right) modes at $m_H = 200$, 400, and 900 GeV. The uncertainties are listed in decreasing order of their effect on the total uncertainty; additional uncertainties with smaller effects are not shown.

As no significant excess is observed, exclusion limits are calculated with a modified frequentist method [103], 822 also known as CL_s , using the \tilde{q}_{μ} test statistic in the asymptotic approximation [104]. The observed lim-823 its can be compared with expectations by generating 'Asimov' data sets, which are representative event 824 samples that provide both the median expectation for an experimental result and its expected statistical 825 variation in the asymptotic approximation, as described in Ref. [104]. When producing the Asimov data 826 set for the expected limits, the background-only hypothesis is assumed and the cross-sections for both 827 ggF and VBF production of the heavy Higgs boson are set to zero. The remaining nuisance parameters 828 are set to the value that maximizes the likelihood function for the observed data (profiled). When using 829 the asymptotic procedure to calculate limits it is necessary to generate an Asimov data set both for the 830 background-only hypothesis and for the signal hypothesis. When setting the observed limits, the cross-831 section for the other production mode not under consideration is profiled to data before generating the 832 background-only Asimov data set. 833

834 11. Results

Limits on the cross-section times branching ratio from the combination of all of the searches are shown in Fig. 12. In the mass range between 140 GeV and 1 TeV, the 95% confidence level (CL) upper limits on the cross-section times branching ratio for heavy Higgs boson production are between 0.19 (0.20) pb and 0.011 (0.013) pb for the ggF (VBF) channels. The excursions into the 2σ band around the expected limit originate from local deviations in the input distributions. For example, the excess occurring around 200 GeV and the deficit occurring around 300 GeV arise from the $\ell\ell\ell\ell\ell$ (see Fig. 1) search. Deficits at higher mass are driven by fluctuations in the $\ell\ell qq$ search (see Figs. 3 and 6).

Figure 13 shows exclusion limits in the $\cos(\beta - \alpha)$ versus $\tan \beta$ plane for Type-I and Type-II 2HDMs, 842 for a heavy Higgs boson with mass $m_H = 200$ GeV. This m_H value is chosen so the assumption of a 843 narrow-width Higgs boson is valid over most of the parameter space, and the experimental sensitivity is 844 at a maximum. As explained in Section 3.2, the range of $\cos(\beta - \alpha)$ and $\tan \beta$ explored is limited to 845 the region where the assumption of a heavy narrow-width Higgs boson with negligible interference is 846 valid. When calculating the limits at a given choice of $\cos(\beta - \alpha)$ and $\tan \beta$, the relative rate of ggF 847 and VBF production in the fit is set according to the prediction of the 2HDM for that parameter choice. 848 Figure 14 shows exclusion limits as a function of the heavy Higgs boson mass m_H and the parameter tan β 849 for $\cos(\beta - \alpha) = -0.1$. The white regions in the exclusion plots indicate regions of parameter space not 850 excluded by the present analysis; in these regions the cross-section predicted by the 2HDM is below the 851 experimental sensitivity. Compared with recent studies of indirect limits [105], the exclusion presented 852 here is considerably more stringent for Type-I with $\cos(\beta - \alpha) < 2$ and $0.5 < \tan \beta < 2$, and for Type-II 853 with $0.5 < \tan \beta < 2$. 854

The previously published ATLAS results using data collected at $\sqrt{s} = 7$ TeV [5–7] assumed a SM Higgs boson with the relative rate of ggF and VBF production fixed to the SM prediction. Thus, they are not directly comparable with the current results, which assume that the heavy Higgs boson has a narrow width but also allow the rates of ggF and VBF production to vary independently. These results are also not directly comparable with the recent results published by the CMS Collaboration [8] for similar reasons.



(a) ggF



(b) VBF

Figure 12: 95% CL upper limits on $\sigma \times BR(H \to ZZ)$ as a function of m_H , resulting from the combination of all of the searches in the (a) ggF and (b) VBF channels. The solid black line and points indicate the observed limit. The dashed black line indicates the expected limit and the bands the 1- σ and 2- σ uncertainty ranges about the expected limit. The dashed coloured lines indicate the expected limits obtained from the individual searches; for the $\ell\ell qq$ and $\nu\nu qq$ searches, only the combination of the two is shown as they share control regions.



Figure 13: 95% CL exclusion contours in the 2HDM (a) Type-I and (b) Type-II models for $m_H = 200$ GeV, shown as a function of the parameters $\cos(\beta - \alpha)$ and $\tan \beta$. The red hashed area shows the observed exclusion, with the solid red line denoting the edge of the excluded region. The dashed blue line represents the expected exclusion contour and the shaded bands the 1- σ and 2- σ uncertainties on the expectation. The vertical axis range is set such that regions where the light Higgs couplings are significantly altered from their SM values are avoided.



(a) Type-I

(b) Type-II

Figure 14: 95% CL exclusion contours in the 2HDM (a) Type-I and (b) Type-II models for $\cos(\beta - \alpha) = -0.1$, shown as a function of the heavy Higgs boson mass m_H and the parameter $\tan \beta$. The shaded area shows the observed exclusion, with the black line denoting the edge of the excluded region. The blue line represents the expected exclusion contour and the shaded bands the 1- σ and 2- σ uncertainties on the expectation. The grey area masks regions where the width of the boson is greater than 0.5% of m_H . For the choice of $\cos(\beta - \alpha) = -0.1$ the light Higgs couplings are not significantly altered from their SM values.

860 12. Summary

A search is presented for a high-mass Higgs boson in the $H \to ZZ \to \ell^+ \ell^- \ell^+ \ell^-$, $H \to ZZ \to \ell^+ \ell^- \nu \bar{\nu}$, 861 $H \to ZZ \to \ell^+ \ell^- q\bar{q}$, and $H \to ZZ \to v\bar{v}q\bar{q}$ decay modes using the ATLAS detector at the CERN 862 Large Hadron Collider. The search uses proton-proton collision data at a centre-of-mass energy of 8 TeV 863 corresponding to an integrated luminosity of 20.3 $\rm fb^{-1}$. The results of the search are interpreted in 864 the scenario of a heavy Higgs boson with a width that is small compared with the experimental mass 865 resolution. The Higgs boson mass range considered extends up to 1 TeV for all four decay modes and down 866 to as low as 140 GeV, depending on the decay mode. No significant excess of events over the Standard 867 Model prediction is found. Limits on production and decay of a heavy Higgs boson to two Z bosons 868 are set separately for gluon-fusion and vector-boson-fusion production modes. For the combination of all 869 decay modes, 95% CL upper limits range from 0.19 pb at $m_H = 140$ GeV to 0.011 pb at $m_H = 1$ TeV 870 for the gluon-fusion production mode, and from 0.20 pb at $m_H = 140$ GeV to 0.013 pb at $m_H = 1$ TeV 871 for the vector-boson-fusion production mode. The results are also interpreted in the context of Type-I and 872 Type-II two-Higgs-doublet models, with exclusion contours given in the $\cos(\beta - \alpha)$ versus tan β and m_H 873 versus tan β planes for $m_H = 200$ GeV. This m_H value is chosen so that the assumption of a narrow-874 width Higgs boson is valid over most of the parameter space, and so that the experimental sensitivity is 875 at a maximum. Compared with recent studies of indirect limits, the two-Higgs-doublet model exclusion 876 presented here is considerably more stringent for Type-I with $\cos(\beta - \alpha) < 2$ and $0.5 < \tan \beta < 2$, and 877 for Type-II with $0.5 < \tan \beta < 2$. 878

Acknowledgements

We thank CERN for the very successful operation of the LHC, as well as the support staff from our institutions without whom ATLAS could not be operated efficiently.

We acknowledge the support of ANPCyT, Argentina; YerPhI, Armenia; ARC, Australia; BMWFW and 882 FWF, Austria; ANAS, Azerbaijan; SSTC, Belarus; CNPq and FAPESP, Brazil; NSERC, NRC and CFI, 883 Canada; CERN; CONICYT, Chile; CAS, MOST and NSFC, China; COLCIENCIAS, Colombia; MSMT 884 CR, MPO CR and VSC CR, Czech Republic; DNRF, DNSRC and Lundbeck Foundation, Denmark; 885 EPLANET, ERC and NSRF, European Union; IN2P3-CNRS, CEA-DSM/IRFU, France; GNSF, Georgia; 886 BMBF, DFG, HGF, MPG and AvH Foundation, Germany; GSRT and NSRF, Greece; RGC, Hong Kong 887 SAR, China; ISF, MINERVA, GIF, I-CORE and Benoziyo Center, Israel; INFN, Italy; MEXT and JSPS. 888 Japan; CNRST, Morocco; FOM and NWO, Netherlands; BRF and RCN, Norway; MNiSW and NCN, 889 Poland; GRICES and FCT, Portugal; MNE/IFA, Romania; MES of Russia and ROSATOM, Russian 890 Federation; JINR; MSTD, Serbia; MSSR, Slovakia; ARRS and MIZŠ, Slovenia; DST/NRF, South Africa; 891 MINECO, Spain; SRC and Wallenberg Foundation, Sweden; SER, SNSF and Cantons of Bern and 892 Geneva, Switzerland; NSC, Taiwan; TAEK, Turkey; STFC, the Royal Society and Leverhulme Trust, 893 United Kingdom; DOE and NSF, United States of America. 894

The crucial computing support from all WLCG partners is acknowledged gratefully, in particular from CERN and the ATLAS Tier-1 facilities at TRIUMF (Canada), NDGF (Denmark, Norway, Sweden),

⁸⁹⁷ CC-IN2P3 (France), KIT/GridKA (Germany), INFN-CNAF (Italy), NL-T1 (Netherlands), PIC (Spain),

ASGC (Taiwan), RAL (UK) and BNL (USA) and in the Tier-2 facilities worldwide.

A. Corrections to MC simulation for the $\ell \ell q q$ search

In order to improve the description of the data in the resolved ggF channel, corrections are applied to the 900 SHERPA Z + jets simulation (prior to the likelihood fit) as a function of the azimuthal angle between the 901 two signal jets, $\Delta \phi_{jj}$, and the transverse momentum of the leptonic Z boson, $p_T^{\ell\ell}$, following Ref. [30]. 902 The simulation does not model well the observed $\Delta \phi_{jj}$ distribution in the untagged control regions for 903 $p_{\rm T}^{\ell\ell}$ < 120 GeV; this is not seen at higher $p_{\rm T}^{\ell\ell}$ or in the tagged control region. In order to improve the 904 modelling, the Z + jj component of the background with $p_T^{\ell\ell} < 120$ GeV is scaled by a linear function derived from the control region with no *b*-tagged jets at low $p_T^{\ell\ell}$ with non-*Z* boson backgrounds subtracted. 905 906 Half the value of the correction is taken as a systematic uncertainty where it is applied. In the Z+hf sample 907 with $p_T^{\ell\ell} < 120$ GeV, the full value of the correction is taken as an uncertainty. For $p_T^{\ell\ell} > 120$ GeV, no 908 correction is applied for any sample. In this region, a linear fit is performed to the data/MC ratio of $\Delta \phi_{ii}$ 909 in the untagged subchannel after subtracting the small non-Z background, and the uncertainty on the fitted 910 slope taken as an uncertainty for all Z + jets samples. Following this correction, the description of the 911 $p_{\rm T}^{\ell\ell}$ distribution in the control region with no *b*-tagged jets also improves, but there is still some residual 912 discrepancy seen in the control regions that have b-tagged jets. Thus, the Z+hf background component 913 is scaled by a function logarithmic in $p_T^{\ell\ell}$, determined from the combination of the control regions with 914 one or more *b*-tagged jets (after subtracting the Z + jj and non-Z + jets background components). An 915 uncertainty of half this correction is applied for all Z + jets channels. (All these uncertainties are taken to 916 be uncorrelated between the Z +light-jet and Z +hf samples.) Following these corrections, the simulation 917 models both the $\Delta \phi_{jj}$ and $p_T^{\ell \ell}$ distributions well in all Z + jets control regions. 918

For the VBF channel, no significant differences are seen in the $\Delta \phi_{jj}$ and $p_T^{\ell \ell}$ distributions, but there is a small difference in the $m_{\ell \ell j j}$ distribution in the control region. The simulated Z + jets background is corrected for this bin-by-bin and the full value of this correction is taken as an uncertainty, again uncorrelated between light- and heavy-flavour samples. No corrections are needed for the merged-jet ggF channel given the small sample size available.

It has been observed in an unfolded measurement of the $p_{\rm T}$ distribution of $t\bar{t}$ quark pairs that the simulation does not accurately describe the $p_{\rm T}^{t\bar{t}}$ distribution [106]. To correct for this, $t\bar{t}$ MC events are weighed by a function of $p_{\rm T}^{t\bar{t}}$ taken from 7 TeV data from Ref. [106] in order to make the simulation match the data. The correction is validated for 8 TeV data using the $e\mu$ top-quark control region, and the uncertainty in this correction is estimated by varying it from 50% to 150% of its nominal value.

B. Flavour tagging in the $\ell \ell q q$ and $\nu \nu q q$ search

In order to constrain the normalisations of the various flavour components of the Z+jets (Z + jj, Z + cj, Z + cj, Z + bj, and Z+hf) and W+jets (W + jj, W + cj and W+hf) backgrounds in the $\ell\ell qq$ and $\nu\nu qq$ channels it is necessary to distinguish the different combinations of jet flavour. This is achived by combining the information from the MV1c *b*-tagging discriminant of the two signal jets.

Four MV1c selection criteria (or operating points) are calibrated, corresponding to average *b*-efficiencies of 80%, 70%, 60% and 50% for *b*-jets with $p_{\rm T} > 20$ GeV, as measured in simulated $t\bar{t}$ events. Based on these, 5 bins in MV1c are defined, which are referred to as:

⁹³⁷ Very Loose (VL): > 80% *b*-tagging efficiency;

- **Loose (L):** 80 70% *b*-tagging efficiency;
- 939 **Medium (M):** 70 60% *b*-tagging efficiency;
- **Tight (T):** 60 50% *b*-tagging efficiency;
- ⁹⁴¹ Very Tight (VT): < 50% *b*-tagging efficiency.

Events are then classified based on the combination of the binned MV1c operating point for the two signal

⁹⁴³ jets, as outlined in Fig. 15.



Figure 15: Event classification as a function of the output of the MV1c *b*-tagging algorithm for the two signal jets. The bin boundaries denote the operating points (MV1c(jet) OP) corresponding to b-tagging efficiencies of 100%, 80%, 70%, 50%, i.e., the *b*-jet purity increases from left (bottom) to right (top). The event categories are labelled as VL, L, M, T and LT according to the definition in the text.

The resulting MV1c event clasiffication is shown in Fig. 16 and Fig. 17 for the $\ell \ell q q Z$ + jets and $\nu \nu q q$

W + jets control regions, respectively, where the date is well described by the MC simulation after the

⁹⁴⁶ combined fit described in Section 10.



Figure 16: The distribution of the MV1c event classification, based on the two signal jets, in the Z + jets control region in the (a) untagged ggF and (b) tagged ggF channels of the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ search. The *b*-jet purity generally increases from left to right. The dashed line shows the total background used as input to the fit. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.



Figure 17: The distribution of the MV1c event classification, based on the two signal jets, in the W + jets (a) 0-*b*-tag and (b) 1-*b*-tag control regions of the $H \rightarrow ZZ \rightarrow v\bar{v}q\bar{q}$ search. The *b*-jet purity generally increases from left to right. The dashed line shows the total background used as input to the fit. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.

947

_

948 **References**

949 950 951	[1]	ATLAS Collaboration, <i>Observation of a new particle in the search for the Standard Model Higgs boson with the ATLAS detector at the LHC</i> , Phys. Lett. B 716 (2012) 1, arXiv: 1207.7214 [hep-ex].
952 953 954	[2]	CMS Collaboration, <i>Observation of a new boson at a mass of 125 GeV with the CMS experiment at the LHC</i> , Phys. Lett. B 716 (2012) 30, arXiv: 1207.7235 [hep-ex].
955 956	[3]	G. Branco et al., <i>Theory and phenomenology of two-Higgs-doublet models</i> , Phys. Rep. 516 (2012) 1–102, arXiv: 1106.0034 [hep-ph].
957 958	[4]	A. Hill and J. van der Bij, <i>Strongly interacting singlet-doublet Higgs model</i> , Phys. Rev. D 36 (1987) 3463–3473.
959 960 961	[5]	ATLAS Collaboration, Search for the Standard Model Higgs boson in the decay channel $H \rightarrow ZZ^{(*)} \rightarrow 4\ell$ with 4.8 fb ⁻¹ of pp collision data at $\sqrt{s} = 7$ TeV with ATLAS, Phys. Lett. B 710 (2012) 383, arXiv: 1202.1415 [hep-ex].
962 963 964	[6]	ATLAS Collaboration, Search for a standard model Higgs boson in the mass range 200–600 GeV in the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ decay channel with the ATLAS detector, Phys. Lett. B 717 (2012) 70, arXiv: 1206.2443 [hep-ex].
965 966 967	[7]	ATLAS Collaboration, Search for a standard model Higgs boson in the $H \rightarrow ZZ \rightarrow \ell^+ \ell^- v \bar{v}$ decay channel using 4.7 fb ⁻¹ of $\sqrt{s} = 7$ TeV data with the ATLAS detector, Phys. Lett. B 717 (2012) 29, arXiv: 1205.6744 [hep-ex].
968 969	[8]	CMS Collaboration, <i>Search for a Higgs boson in the mass range from 145 to 1000 GeV decaying to a pair of W or Z bosons</i> (2015), submitted to JHEP, arXiv: 1504.00936 [hep-ex].
970 971	[9]	ATLAS Collaboration, <i>The ATLAS Experiment at the CERN Large Hadron Collider</i> , JINST 3 (2008) S08003.
972 973	[10]	ATLAS Collaboration, <i>Performance of the ATLAS Trigger System in 2010</i> , Eur. Phys. J. C 72 (2012) 1849, arXiv: 1110.1530 [hep-ex].
974 975 976	[11]	S. Alioli et al., <i>NLO Higgs boson production via gluon fusion matched with shower in POWHEG</i> , JHEP 0904 (2009) 002, arXiv: 0812.0578 [hep-ph].
977 978 979	[12]	P. Nason and C. Oleari, <i>NLO Higgs boson production via vector-boson fusion matched with shower in POWHEG</i> , JHEP 1002 (2010) 037, arXiv: 0911.5299 [hep-ph].
980 981	[13]	ATLAS Collaboration, <i>ATLAS tunes of Pythia 6 and Pythia 8 for MC11</i> , ATL-PHYS-PUB-2011-009, 2011, url: http://cds.cern.ch/record/1363300.
982 983	[14]	ATLAS Collaboration, <i>New ATLAS event generator tunes to 2010 data</i> , ATL-PHYS-PUB-2011-008, 2011, URL: http://cds.cern.ch/record/1345343.
984 985	[15]	HL. Lai et al., <i>New parton distributions for collider physics</i> , Phys. Rev. D 82 (2010) 074024, arXiv: 1007.2241 [hep-ph].

986 987 988	[16]	J. Pumplin et al., New generation of parton distributions with uncertainties from global QCD analysis, JHEP 0207 (2002) 012, arXiv: hep-ph/0201195 [hep-ph].
989 990 991	[17]	R. V. Harlander, S. Liebler and H. Mantler, <i>SusHi: A program for the calculation of Higgs production in gluon fusion and bottom-quark annihilation in the Standard Model and the MSSM</i> , Comp. Phys. Comm. 184 (2013) 1605–1617, arXiv: 1212.3249 [hep-ph].
992 993 994	[18]	R. V. Harlander and W. B. Kilgore, Next-to-next-to-leading order Higgs production at hadron colliders, Phys. Rev. Lett. 88 (2002) 201801, arXiv: hep-ph/0201206 [hep-ph].
995 996 997	[19]	R. V. Harlander and W. B. Kilgore, <i>Higgs boson production in bottom quark fusion at next-to-next-to leading order</i> , Phys. Rev. D 68 (2003) 013001, arXiv: hep-ph/0304035 [hep-ph].
998 999	[20]	U. Aglietti et al., <i>Two-loop light fermion contribution to Higgs production and decays</i> , Phys. Lett. B 595 (2004) 432–441, arXiv: hep-ph/0404071.
1000 1001 1002	[21]	R. Bonciani, G. Degrassi and A. Vicini, On the Generalized Harmonic Polylogarithms of One Complex Variable, Comp. Phys. Comm. 182 (2011) 1253–1264, arXiv: 1007.1891 [hep-ph].
1003 1004 1005	[22]	R. Harlander and P. Kant, <i>Higgs production and decay: Analytic results at next-to-leading order QCD</i> , JHEP 0512 (2005) 015, arXiv: hep-ph/0509189 [hep-ph].
1006 1007 1008	[23]	D. Eriksson, J. Rathsman and O. Stal, 2HDMC: Two-Higgs-Doublet Model Calculator Physics and Manual, Comp. Phys. Comm. 181 (2010) 189–205, arXiv: 0902.0851 [hep-ph].
1009 1010	[24]	T. Gleisberg et al., <i>Event generation with SHERPA 1.1</i> , JHEP 0902 (2009) 007, arXiv: 0811.4622 [hep-ph].
1011 1012 1013	[25]	M. L. Mangano et al., ALPGEN, a generator for hard multiparton processes in hadronic collisions, JHEP 0307 (2003) 001, arXiv: hep-ph/0206293.
1014 1015 1016	[26]	M. L. Mangano et al., Matching matrix elements and shower evolution for top-quark production in hadronic collisions, JHEP 0701 (2007) 013, arXiv: hep-ph/0611129 [hep-ph].
1017 1018 1019	[27]	M. L. Ciccolini, S. Dittmaier and M. Kramer, Electroweak radiative corrections to associated WH and ZH production at hadron colliders, Phys. Rev. D 68 (2003) 073003, arXiv: hep-ph/0306234.
1020 1021	[28]	A. Denner et al., <i>EW corrections to Higgs strahlung at the Tevatron and the LHC with HAWK</i> , PoS EPS-HEP2011 (2011) 235, arXiv: 1112.5258 [hep-ph].
1022 1023 1024	[29]	A. Djouadi, J. Kalinowski and M. Spira, <i>HDECAY: A program for Higgs boson decays in the standard model and its supersymmetric extension</i> , Comp. Phys. Comm. 108 (1998) 56–74, arXiv: hep-ph/9704448.
1025 1026 1027	[30]	ATLAS Collaboration, Search for the $b\bar{b}$ decay of the Standard Model Higgs boson in associated $(W/Z)H$ production with the ATLAS detector, JHEP 1501 (2015) 069, arXiv: 1409.6212 [hep-ex].

1028 1029 1030	[31]	Cascioli, F. and <i>et al.</i> , <i>ZZ production at hadron colliders in NNLO QCD</i> , Phys. Lett. B 735 (2014) 311, extended by the authors to provide NNLO/NLO <i>k</i> -factors as a function of m_{ZZ} , arXiv: 1405.2219 [hep-ph].
1031 1032 1033	[32]	A. Bierweiler, T. Kasprzik and J. H. Kühn, <i>Vector-boson pair production at the LHC to</i> $O(\alpha^3)$ <i>accuracy</i> , JHEP 1312 (2013) 071, arXiv: 1305.5402 [hep-ph].
1034 1035 1036	[33]	J. Baglio, L. D. Ninh and M. M. Weber, <i>Massive gauge boson pair production at the LHC: a next-to-leading order story</i> , Phys. Rev. D 88 (2013) 113005, arXiv: 1307.4331.
1037 1038 1039	[34]	S. Gieseke, T. Kasprzik and J. H. Kühn, Vector-boson pair production and electroweak corrections in HERWIG++, Eur. Phys. J. C 74 (2014) 2988.
1040 1041	[35]	J. M. Campbell, R. K. Ellis and C. Williams, <i>Vector boson pair production at the LHC</i> , JHEP 1107 (2011) 018, arXiv: 1105.0020 [hep-ph].
1042 1043 1044	[36]	N. Kauer and G. Passarino, Inadequacy of zero-width approximation for a light Higgs boson signal, JHEP 1208 (2012) 116, arXiv: 1206.4803 [hep-ph].
1045 1046	[37]	N. Kauer, Interference effects for $H \to WW/ZZ \to \ell \bar{\nu}_{\ell} \bar{\ell} \nu_{\ell}$ searches in gluon fusion at the LHC, JHEP 1312 (2013) 082, arXiv: 1310.7011 [hep-ph].
1047	[38]	G. Passarino, <i>Higgs CAT</i> , Eur. Phys. J. C 74 (2014) 2866, arXiv: 1312.2397 [hep-ph].
1048 1049 1050	[39]	S. Frixione, P. Nason and B. R. Webber, <i>Matching NLO QCD and parton showers in heavy flavour production</i> , JHEP 0308 (2003) 007, arXiv: hep-ph/0305252.
1051 1052 1053	[40]	G. Corcella et al., <i>HERWIG 6.5: an event generator for Hadron Emission Reactions With Interfering Gluons (including supersymmetric processes)</i> , JHEP 0101 (2001) 010, arXiv: hep-ph/0011363.
1054 1055 1056	[41]	J. Butterworth, J. R. Forshaw and M. Seymour, <i>Multiparton interactions in photoproduction at HERA</i> , Z. Phys. C 72 (1996) 637–646, arXiv: hep-ph/9601371 [hep-ph].
1057 1058	[42]	T. Sjostrand, S. Mrenna and P. Z. Skands, <i>PYTHIA 6.4 Physics and Manual</i> , JHEP 0605 (2006) 026, arXiv: hep-ph/0603175.
1059 1060	[43]	J. Alwall et al., <i>MadGraph 5 : Going Beyond</i> , JHEP 1106 (2011) 128, arXiv: 1106.0522 [hep-ph].
1061 1062 1063	[44]	B. P. Kersevan and E. Richter-Was, <i>The Monte Carlo event generator AcerMC version 2.0 with interfaces to PYTHIA 6.2 and HERWIG 6.5</i> , Comp. Phys. Comm. 184 (2004) 919–985, arXiv: hep-ph/0405247 [hep-ph].
1064 1065	[45]	T. Sjostrand, S. Mrenna and P. Skands, <i>A Brief Introduction to PYTHIA 8.1</i> , Comp. Phys. Comm. 178 (2008) 852, arXiv: 0710.3820 [hep-ph].
1066 1067 1068	[46]	J. M. Campbell, R. K. Ellis and C. Williams, Bounding the Higgs width at the LHC using full analytic results for $gg - > e^-e^+\mu^-\mu^+$, JHEP 1404 (2014) 060, arXiv: 1311.3589 [hep-ph].

1069 1070 1071	[47]	K. Melnikov and F. Petriello, Electroweak gauge boson production at hadron colliders through $O(\alpha_s^2)$, Phys. Rev. D 74 (2006) 114017, arXiv: hep-ph/0609070.
1072 1073	[48]	ATLAS Collaboration, <i>Further ATLAS tunes of Pythia 6 and Pythia 8</i> , ATL-PHYS-PUB-2011-014, 2011, URL: http://cds.cern.ch/record/1400677.
1074 1075 1076	[49]	S. Catani et al., Vector boson production at hadron colliders: a fully exclusive QCD calculation at NNLO, Phys. Rev. Lett. 103 (2009) 082001, arXiv: 0903.2120 [hep-ph].
1077 1078 1079	[50]	S. Catani and M. Grazzini, <i>An NNLO subtraction formalism in hadron collisions and its application to Higgs boson production at the LHC</i> , Phys. Rev. Lett. 98 (2007) 222002, arXiv: hep-ph/0703012 [hep-ph].
1080 1081	[51]	P. Nason, <i>A new method for combining NLO QCD with shower Monte Carlo algorithms</i> , JHEP 0411 (2004) 040, arXiv: hep-ph/0409146.
1082 1083 1084	[52]	S. Frixione, P. Nason and C. Oleari, Matching NLO QCD computations with Parton Shower simulations: the POWHEG method, JHEP 0711 (2007) 070, arXiv: 0709.2092 [hep-ph].
1085 1086	[53]	S. Alioli et al., <i>A general framework for implementing NLO calculations in shower Monte Carlo programs: the POWHEG BOX</i> , JHEP 1006 (2010) 043, arXiv: 1002.2581 [hep-ph].
1087 1088 1089	[54]	B. Cooper et al., <i>Importance of a consistent choice of alpha(s) in the matching of AlpGen and Pythia</i> , Eur. Phys. J. C 72 (2012) 2078, arXiv: 1109 .5295 [hep-ph].
1090 1091 1092	[55]	M. Czakon, P. Fiedler and A. Mitov, <i>Total Top-Quark Pair-Production Cross Section at Hadron Colliders Through</i> $O(\alpha_s^4)$, Phys. Rev. Lett. 110 (2013) 252004, arXiv: 1303.6524 [hep-ph].
1093 1094 1095	[56]	M. Czakon and A. Mitov, <i>Top++: A Program for the Calculation of the Top-Pair Cross-Section at Hadron Colliders</i> , Comp. Phys. Comm. 185 (2014) 2930, arXiv: 1112.5675 [hep-ph].
1096 1097	[57]	N. Kidonakis, <i>NNLL resummation for s-channel single top quark production</i> , Phys. Rev. D 81 (2010) 054028, arXiv: 1001.5034 [hep-ph].
1098 1099 1100	[58]	N. Kidonakis, <i>Two-loop soft anomalous dimensions for single top quark associated production with a W- or H-</i> , Phys. Rev. D 82 (2010) 054018, arXiv: 1005.4451 [hep-ph].
1101 1102	[59]	N. Kidonakis, <i>Next-to-next-to-leading-order collinear and soft gluon corrections for t-channel single top quark production</i> , Phys. Rev. D 83 (2011) 091503, arXiv: 1103.2792 [hep-ph].
1103 1104	[60]	P. Nason and G. Zanderighi, W^+W^- , WZ and ZZ production in the POWHEG-BOX-V2, Eur. Phys. J. C 74 (2014) 2702, arXiv: 1311.1365 [hep-ph].
1105 1106	[61]	J. M. Campbell and R. K. Ellis, <i>An update on vector boson pair production at hadron colliders</i> , Phys. Rev. D 60 (1999) 113006, arXiv: hep-ph/9905386 [hep-ph].
1107 1108	[62]	J. Ohnemus and W. J. Stirling, Order alpha-s corrections to the differential cross-section for the W H intermediate mass Higgs signal, Phys. Rev. D 47 (1993) 2722–2729.
1109 1110	[63]	H. Baer, B. Bailey and J. Owens, O (alpha-s) Monte Carlo approach to W + Higgs associated production at hadron supercolliders, Phys. Rev. D 47 (1993) 2730–2734.

O. Brein, A. Djouadi and R. Harlander, [64] 1111 NNLO QCD corrections to the Higgs-strahlung processes at hadron colliders, 1112 Phys. Lett. B 579 (2004) 149–156, arXiv: hep-ph/0307206 [hep-ph]. 1113 L. Altenkamp et al., Gluon-induced Higgs-strahlung at next-to-leading order QCD, [65] 1114 JHEP 1302 (2013) 078, arXiv: 1211.5015 [hep-ph]. 1115 ATLAS Collaboration, The ATLAS Simulation Infrastructure, Eur. Phys. J. C 70 (2010) 823, [66] 1116 arXiv: 1005.4568 [hep-ex]. 1117 [67] S. Agostinelli et al., GEANT4: A Simulation toolkit, 1118 Nucl. Instrum. Meth. A 506 (2003) 250-303. 1119 ATLAS Collaboration, The simulation principle and performance of the ATLAS fast calorimeter [68] 1120 simulation FastCaloSim, ATL-PHYS-PUB-2010-013, 2010, 1121 URL: http://cds.cern.ch/record/1300517. 1122 [69] A. Sherstnev and R. Thorne, Parton Distributions for LO Generators, 1123 Eur. Phys. J. C 55 (2008) 553–575, arXiv: 0711.2473 [hep-ph]. 1124 [70] ATLAS Collaboration, Measurement of the muon reconstruction performance of the ATLAS 1125 detector using 2011 and 2012 LHC proton–proton collision data, Eur. Phys. J. C 74 (2014) 3130, 1126 arXiv: 1407.3935 [hep-ex]. 1127 [71] ATLAS Collaboration, Electron reconstruction and identification efficiency measurements with 1128 the ATLAS detector using the 2011 LHC proton-proton collision data, 1129 Eur. Phys. J. C 74 (2014) 2941, arXiv: 1404.2240 [hep-ex]. 1130 [72] ATLAS Collaboration, 1131 Electron and photon energy calibration with the ATLAS detector using LHC Run 1 data, 1132 Eur. Phys. J. C 74 (2014) 3071, arXiv: 1407.5063 [hep-ex]. 1133 ATLAS Collaboration, Electron efficiency measurements with the ATLAS detector using the [73] 1134 2012 LHC proton–proton collision data, ATLAS-CONF-2014-032, 2014, 1135 url: http://cdsweb.cern.ch/record/1706245. 1136 M. Cacciari, G. P. Salam and G. Soyez, FastJet User Manual, Eur. Phys. J. C 72 (2012) 1896, [74] 1137 arXiv: 1111.6097 [hep-ph]. 1138 [75] M. Cacciari, G. P. Salam and G. Soyez, The anti- k_t jet clustering algorithm, 1139 JHEP 0804 (2008) 063, arXiv: 0802.1189 [hep-ph]. 1140 [76] ATLAS Collaboration, Jet energy measurement and its systematic uncertainty in proton-proton 1141 collisions at $\sqrt{s} = 7$ TeV with the ATLAS detector, Eur. Phys. J. C 75 (2015) 17, 1142 arXiv: 1406.0076 [hep-ex]. 1143 [77] ATLAS Collaboration, 1144 *Pile-up corrections for jets from proton–proton collisions at* $\sqrt{s} = 7$ *TeV in ATLAS in 2011,* 1145 ATLAS-CONF-2012-064, 2012, url: http://cdsweb.cern.ch/record/1459529. 1146 ATLAS Collaboration, Pile-up subtraction and suppression for jets in ATLAS, [78] 1147 ATLAS-CONF-2013-083, 2013, uRL: http://cdsweb.cern.ch/record/1570994. 1148 [79] ATLAS Collaboration, Performance of missing transverse momentum reconstruction in 1149 proton–proton collisions at $\sqrt{s} = 7$ TeV with ATLAS, Eur. Phys. J. C 72 (2012) 1844, 1150 arXiv: 1108.5602 [hep-ex]. 1151

1152 1153 1154	[80]	ATLAS Collaboration, <i>Commissioning of the ATLAS high performance b-tagging algorithms in the</i> 7 <i>TeV collision data</i> , ATLAS-CONF-2011-102, 2011, URL: http://cdsweb.cern.ch/record/1369219.
1155 1156 1157	[81]	ATLAS Collaboration, <i>Measurement of the b-tag Efficiency in a Sample of Jets Containing</i> <i>Muons with</i> 5 fb^{-1} <i>of data from the ATLAS detector</i> , ATLAS-CONF-2012-043, 2012, URL: http://cdsweb.cern.ch/record/1435197.
1158 1159 1160	[82]	ATLAS Collaboration, <i>Calibration of b-tagging using dileptonic top pair events in a combinatorial likelihood approach with the ATLAS experiment</i> , ATLAS-CONF-2014-004, 2014, URL: http://cdsweb.cern.ch/record/1664335.
1161 1162 1163	[83]	ATLAS Collaboration, <i>Measurement of the Mistag Rate of b-tagging algorithms with</i> $5 fb^{-1}$ <i>of Data Collected by the ATLAS Detector</i> , ATLAS-CONF-2012-040, 2012, URL: http://cdsweb.cern.ch/record/1435194.
1164 1165 1166 1167	[84]	ATLAS Collaboration, <i>Measurements of Higgs boson production and couplings in the four-lepton channel in pp</i> <i>collisions at center-of-mass energies of 7 and 8 TeV with the ATLAS detector</i> , Phys. Rev. D 91 (2015) 012006, arXiv: 1408.5191 [hep-ex].
1168 1169 1170	[85]	ATLAS Collaboration, Search for Invisible Decays of a Higgs Boson Produced in Association with a Z Boson in ATLAS, Phys. Rev. Lett. 112 (2014) 201802, arXiv: 1402.3244 [hep-ex].
1171 1172 1173 1174	[86]	I. W. Stewart and F. J. Tackmann, <i>Theory uncertainties for Higgs mass and other searches using jet bins</i> , Phys. Rev. D 85 (2012) 034011, URL: http://link.aps.org/doi/10.1103/PhysRevD.85.034011.
1175 1176 1177 1178	[87]	S. Gangal and F. J. Tackmann, Next-to-leading-order uncertainties in Higgs + jets from gluon fusion, Phys. Rev. D 87 (2013) 093008, URL: http://link.aps.org/doi/10.1103/PhysRevD.87.093008.
1179 1180	[88]	ATLAS Collaboration, <i>Improved luminosity determination in pp collisions at</i> $\sqrt{s} = 7$ <i>TeV using the ATLAS detector at the LHC</i> , Eur. Phys. J. C 73 (2013) 2518, arXiv: 1302.4393 [hep-ex].
1181 1182 1183	[89]	ATLAS Collaboration, Jet energy measurement with the ATLAS detector in proton–proton collisions at $\sqrt{s} = 7$ TeV, Eur. Phys. J. C 73 (2013) 2304, arXiv: 1112.6426 [hep-ex].
1184 1185	[90]	ATLAS Collaboration, <i>Jet energy resolution in proton–proton collisions at</i> $\sqrt{s} = 7$ <i>TeV recorded in 2010 with the ATLAS detector</i> , Eur. Phys. J. C 73 (2013) 2306, arXiv: 1210.6210 [hep-ex].
1186 1187 1188	[91]	ATLAS Collaboration, Performance of Missing Transverse Momentum Reconstruction in ATLAS studied in Proton–Proton Collisions recorded in 2012 at $\sqrt{s} = 8$ TeV, ATLAS-CONF-2013-082, 2013, URL: http://cdsweb.cern.ch/record/1570993.
1189 1190 1191	[92]	ATLAS Collaboration, <i>Performance of jet substructure techniques for large-R jets in proton–proton collisions at</i> $\sqrt{s} = 7$ <i>TeV using the ATLAS detector</i> , JHEP 1309 (2013) 076, arXiv: 1306.4945 [hep-ex].
1192 1193 1194	[93]	LHC Higgs Cross Section Working Group, Handbook of LHC Higgs Cross Sections: 2. Differential distributions, 2012, arXiv: 1201.3084 [hep-ph].

1195 1196	[94]	M. Botje et al., <i>The PDF4LHC working group interim recommendations</i> , 2011, arXiv: 1101.0538 [hep-ph].
1197 1198	[95]	S. Alekhin et al., <i>The PDF4LHC working group interim report</i> , 2011, arXiv: 1101.0536 [hep-ph].
1199 1200 1201	[96]	ATLAS Collaboration, <i>Determination of the off-shell Higgs boson signal strength in the high-mass ZZ and WW final states with the ATLAS detector</i> (2015), submitted to Eur. Phys. J. C, arXiv: 1503.01060 [hep-ex].
1202 1203 1204	[97]	M. Bonvini et al., Signal-background interference effects for $gg \rightarrow H \rightarrow W^+W^-$ beyond leading order, Phys. Rev. D 88 (2013) 034032, arXiv: 1304.3053 [hep-ph].
1205 1206 1207	[98]	ATLAS Collaboration, <i>Combined search for the Standard Model Higgs boson in pp collisions at</i> $\sqrt{s} = 7$ <i>TeV with the ATLAS detector</i> , Phys. Rev. D 86 (2012) 032003, arXiv: 1207.0319 [hep-ex].
1208 1209 1210	[99]	ATLAS Collaboration, <i>Procedure for the LHC Higgs boson search combination in summer 2011</i> , ATL-PHYS-PUB-2011-011, 2011, URL: http://cds.cern.ch/record/1375842.
1211 1212	[100]	L. Moneta et al., <i>The RooStats Project</i> , PoS ACAT2010 (2010) 057, arXiv: 1009.1003 [physics.data-an].
1213 1214 1215	[101]	K. Cranmer et al., HistFactory: A tool for creating statistical models for use with RooFit and RooStats, CERN-OPEN-2012-016 (2012), URL: http://cdsweb.cern.ch/record/1456844.
1216 1217 1218	[102]	W. Verkerke et al., <i>The RooFit Toolkit for Data Modelling</i> , Available from http://roofit.sourceforge.net or with recent versions of the ROOT framework available at http://root.cern.ch.
1219 1220	[103]	A. L. Read, <i>Presentation of search results: The CL(s) technique</i> , J. Phys. G 28 (2002) 2693–2704.
1221 1222	[104]	G. Cowan et al., <i>Asymptotic formulae for likelihood-based tests of new physics</i> , Eur. Phys. J. C 71 (2011) 1554, arXiv: 1007.1727 [physics.data-an].
1223 1224	[105]	B. Dumont et al., Addendum to "Constraints on and future prospects for Two-Higgs-Doublet Models in light of the LHC Higgs signal" (2014), arXiv: 1409.4088 [hep-ph].
1225 1226 1227	[106]	ATLAS Collaboration, <i>Measurements of normalized differential cross-sections for t</i> \overline{t} <i>production in pp collisions at</i> $\sqrt{s} = 7$ <i>TeV using the ATLAS detector</i> , Phys. Rev. D 90 (2014) 072004, arXiv: 1407.0371 [hep-ex].

1228 Auxiliary Material

Component	Subchannel					
		ggr		V DI	V 11	
$qq \rightarrow ZZ$	50	±	20	5 ± 4	2 ± 1	
$gg \rightarrow ZZ$	280	±	20	15 ± 2	7 ± 1	
Reducible $(Z + jets, top)$	12	±	3	1 ± 1	0.5 ± 0.3	
SM Background	340	±	20	21 ± 4	9 ± 1	
Data		316		22	9	
$H \to ZZ \to \ell^+ \ell^- q\bar{q} (1 \text{ pb})$						
$ggF(m_H = 400 \text{ GeV})$	11	±	1	2.3 ± 0.3	0.8 ± 0.1	
VBF ($m_H = 400 \text{ GeV}$)	6	±	1	8 ± 1	0.31 ± 0.04	
$ggF(m_H = 900 \text{ GeV})$	12	±	1	5 ± 1	1.6 ± 0.2	
$VBF (m_H = 900 \text{ GeV})$	8	±	1	10 ± 1	0.32 ± 0.03	

Table 5: Number of selected data events compared to the fitted background predictions for the ATLAS $H \rightarrow ZZ \rightarrow \ell^+ \ell^- \ell^+ \ell^-$ search in the ggF, VBH, and VH channels. Also shown are the signal predictions for $m_H = 400$ GeV and $m_H = 900$ GeV, normalized to $\sigma \times BR = 1$ pb.

Commonweat	Subchannel				
Component	Untagged	Tagged	Merged-jet	VBF	
Z + jj	35300 ± 700	14 ± 3			
Z + cj	5210 ± 730	34 ± 3			
Z + bj	2310 ± 110	59 ± 6			
Z + hf	1610 ± 130	1100 ± 30			
Z + jets	—		62 ± 6	600 ± 30	
tī/Wt	332 ± 14	200 ± 9	0.32 ± 0.05	34 ± 4	
Diboson	1040 ± 70	140 ± 10	5.0 ± 0.5	18 ± 4	
Multijet	152 ± 1	9 ± 5			
$Zh \rightarrow \ell\ell bb$	10.4 ± 0.3	9 ± 4	_	—	
SM background	46000 ± 210	1600 ± 30	67 ± 6	650 ± 30	
Data	46014	1542	73	644	
$H \to ZZ \to \ell^+ \ell^- q\bar{q} \ (1 \text{ pb})$					
$ggF(m_H = 400 \text{ GeV})$	251 ± 10	71 ± 4		$0. \pm 0.2$	
VBF ($m_H = 400 \text{ GeV}$)	1.3 ± 4.5	0.5 ± 2		120 ± 10	
$ggF(m_H = 900 \text{ GeV})$	202 ± 16	70 ± 7	160 ± 30	2 ± 4	
$\text{VBF}\left(m_{H}=900\text{ GeV}\right)$	1.9 ± 4.2	5 ± 10	17 ± 40	100 ± 9	

Table 6: Number of selected data events compared to the fitted background predictions for the ATLAS $H \rightarrow ZZ \rightarrow \ell^+ \ell^- q\bar{q}$ search in the untagged (< 2 *b*-tagged jets), tagged (= 2 *b*-tagged jets), and merged-jet ggF subchannels, along with the VBF subchannel. Also shown are the signal predictions for $m_H = 400$ GeV and $m_H = 900$ GeV, normalized to $\sigma \times BR = 1$ pb.

DKAFI

	Mass and subchannel					
Component	$m_H = 400 \text{ GeV}$	selection	$m_H = 900 { m Ge}$	eV selection		
	Untagged	Tagged	Untagged	Tagged		
Z + jj	12400 ± 400	1.7 ± 0.3	1590 ± 70	0.26 ± 0.06		
Z + cj	1800 ± 300	3.8 ± 0.4	250 ± 40	0.65 ± 0.08		
Z + bj	790 ± 50	5.1 ± 0.7	121 ± 8	0.9 ± 0.2		
Z + hf	580 ± 50	120 ± 7	120 ± 10	25 ± 2		
W + l	7800 ± 300	1.6 ± 0.1	990 ± 50	0.19 ± 0.03		
W + cl	1200 ± 100	2.9 ± 0.4	160 ± 20	0.44 ± 0.05		
W + hf	450 ± 100	40 ± 10	76 ± 20	9 ± 3		
tī/Wt	2100 ± 100	90 ± 6	520 ± 50	16 ± 1		
Diboson	1200 ± 200	44 ± 5	270 ± 40	14 ± 2		
$Zh \rightarrow \nu\nu bb$	6.8 ± 0.2	5 ± 2	1.4 ± 0.04	1.1 ± 0.5		
SM Background	28400 ± 300	310 ±10	4090 \pm 70	66 ± 4		
Data	28573	323	4096	69		
$H \rightarrow ZZ \rightarrow v\bar{v}q\bar{q} (1 \text{ pb})$						
ggF	320 ± 30	30 ± 3	540 ± 20	75 ± 7		
VBF	2 ± 8	2 ± 8	60 ± 200	8 ± 20		

Table 7: Number of selected data events compared to the fitted background predictions for the ATLAS $H \rightarrow ZZ \rightarrow v\bar{v}q\bar{q}$ search in the untagged (< 2 *b*-tagged jets) and tagged (= 2 *b*-tagged jets) ggF subchannels for the $m_H = 400$ GeV and $m_H = 900$ GeV selection. Also shown are the signal predictions for the corresponding Higgs boson mass normalized to $\sigma \times BR = 1$ pb.



Figure 18: Results from the ATLAS search for a heavy, narrow Higgs boson state decaying to two Z bosons, where each Z boson decays into a pair of either electrons or muons. The solid curve shows the observed 95% CL limits on $\sigma \times BR(H \rightarrow ZZ)$. The dashed curve shows the expected limit and the coloured bands the 1- and 2- σ ranges around the expected limit. (a) ggF mode. (b) VBF mode.



Figure 19: Results from the ATLAS search for a heavy, narrow Higgs boson state decaying to two Z bosons, where one Z boson decays into a pair of either electrons or muons and the other decays into a pair of neutrinos. The solid curve shows the observed 95% CL limits on $\sigma \times BR(H \rightarrow ZZ)$. The dashed curve shows the expected limit and the coloured bands the 1- and 2- σ ranges around the expected limit. (a) ggF mode. (b) VBF mode.



Figure 20: Results from the ATLAS search for a heavy, narrow Higgs boson state decaying to two Z bosons. These plots show the combination of two searches, one in which one Z boson decays to a pair of either electrons or muons and the other to jets, and one in which one Z boson decays to a pair of neutrinos instead of electrons or muons. The solid curve shows the observed 95% CL limits on $\sigma \times BR(H \rightarrow ZZ)$. The dashed curve shows the expected limit and the colored bands the 1- and 2- σ ranges around the expected limit. (a) ggF mode. (b) VBF mode.



Figure 21: The distributions of m_T , the transverse mass of the $Z(\nu\nu)Z(jj)$ system, used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \nu \bar{\nu} q \bar{q}$ search in the (a) untagged and (b) tagged channels, for a Higgs boson mass hypothesis of $m_H = 500$ GeV with the 500 GeV signal region selection. The dashed line shows the total background used as input to the fit. The signal is normalized to a cross-section corresponding to twenty times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.



Figure 22: The distributions of m_T , the transverse mass of the $Z(\nu\nu)Z(jj)$ system, used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \nu \bar{\nu} q \bar{q}$ search in the (a) untagged and (b) tagged channels, for a Higgs boson mass hypothesis of $m_H = 600$ GeV with the 600 GeV signal region selection. The dashed line shows the total background used as input to the fit. The signal is normalized to a cross-section corresponding to twenty times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.



Figure 23: The distributions of m_T , the transverse mass of the $Z(\nu\nu)Z(jj)$ system, used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \nu \bar{\nu} q \bar{q}$ search in the (a) untagged and (b) tagged channels, for a Higgs boson mass hypothesis of $m_H = 700$ GeV with the 700 GeV signal region selection. The dashed line shows the total background used as input to the fit. The signal is normalized to a cross-section corresponding to thirty times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.



Figure 24: The distributions of m_T , the transverse mass of the $Z(\nu\nu)Z(jj)$ system, used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \nu \bar{\nu} q \bar{q}$ search in the (a) untagged and (b) tagged channels, for a Higgs boson mass hypothesis of $m_H = 800$ GeV with the 800 GeV signal region selection. The dashed line shows the total background used as input to the fit. The signal is normalized to a cross-section corresponding to thirty times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.



Figure 25: The distributions of m_T , the transverse mass of the $Z(\nu\nu)Z(jj)$ system, used in the likelihood fit for the $H \rightarrow ZZ \rightarrow \nu \bar{\nu} q \bar{q}$ search in the (a) untagged and (b) tagged channels, for a Higgs boson mass hypothesis of $m_H = 1000$ GeV with the 1000 GeV signal region selection. The dashed line shows the total background used as input to the fit. The signal is normalized to a cross-section corresponding to thirty times the observed limit given in Section 11. The contribution labelled as 'Top' includes both the $t\bar{t}$ and single-top processes. The bottom panes show the ratio of the observed data to the predicted background.

DRAFT

m_H [GeV]	Observed [fb]	Expected [fb]	+1 <i>σ</i> [fb]	+2 <i>σ</i> [fb]	-1σ [fb]	-2σ [fb]
140	190	210	320	460	150	120
145	180	220	320	470	160	120
150	240	200	290	420	140	110
155	240	180	250	370	130	97
160	140	180	260	380	130	96
165	320	180	270	390	130	95
170	220	180	250	370	130	94
175	170	190	270	400	140	100
180	170	220	330	470	160	120
185	220	270	380	540	190	140
190	510	300	430	600	220	160
195	530	320	450	630	230	170
200	360	320	460	650	230	170
220	360	300	420	590	210	160
240	250	260	380	530	190	140
260	340	230	320	460	160	120
280	230	200	280	390	140	100
300	260	170	250	350	130	93
320	93	150	220	300	110	81
340	110	130	190	260	94	70
360	99	120	170	230	84	63
380	120	100	140	200	72	54
400	65	85	120	170	61	45
420	61	72	100	140	52	39
440	71	65	93	130	47	35
460	48	58	82	120	42	31
480	42	53	75	110	38	28
500	44	47	67	94	34	25
520	39	43	61	86	31	23
540	32	39	56	79	28	21
560	30	37	52	74	26	20
580	25	34	49	69	24	18
600	22	32	46	65	23	17
650	26	27	39	55	19	14
700	20	23	32	46	16	12
750	12	19	28	40	14	10
800	12	17	25	36	12	9.2
850	11	15	22	32	11	8.2
900	14	14	20	29	9.9	7.3
950	7.6	13	18	27	9.1	6.8
1 000	11	12	17	25	8.4	6.3

Table 8: Results from the ATLAS search for a heavy, narrow Higgs boson state decaying to two Z bosons, for the ggF production mode. Decay modes considered include $\ell\ell\ell\ell$, $\ell\ell\nu\nu$, $\ell\ell qq$, and $\nu\nu qq$. Tabulated as a function of m_H are the observed 95% CL limits on $\sigma \times BR(H \to ZZ)$, the expected limits, and the 1- and 2- σ ranges around the expected limits.

DRAFT

<i>m_H</i> [GeV]	Observed [fb]	Expected [fb]	+1 <i>σ</i> [fb]	+2 <i>σ</i> [fb]	-1σ [fb]	-2σ [fb]
140	200	160	230	350	110	84
145	270	150	230	360	110	82
150	260	140	210	320	100	76
155	160	130	200	300	96	72
160	99	130	200	310	97	72
165	130	130	200	310	96	71
170	120	130	190	300	94	70
175	95	130	200	300	97	72
180	99	140	220	330	100	77
185	140	160	230	350	110	84
190	220	170	250	370	120	89
195	310	170	260	390	130	94
200	210	140	200	290	97	72
220	160	130	190	280	94	70
240	160	120	180	270	89	67
260	160	110	160	240	81	60
280	100	100	150	220	75	56
300	100	92	130	190	66	49
320	58	80	120	170	58	43
340	42	64	88	130	46	34
360	76	63	91	130	45	34
380	88	55	79	110	40	29
400	65	56	81	120	40	30
420	63	47	67	96	34	25
440	62	44	64	92	32	24
460	48	40	58	84	29	22
480	43	38	56	80	28	21
500	37	33	47	68	24	18
520	31	28	40	57	20	15
540	29	26	38	54	19	14
560	27	25	37	52	18	14
580	25	25	36	52	18	13
600	21	24	35	50	18	13
650	18	20	28	40	14	10
700	16	18	26	37	13	9.7
750	11	16	24	34	12	8.8
800	11	15	22	32	11	8.2
850	10	13	19	28	9.7	7.2
900	14	13	19	27	9.3	6.9
950	8.8	12	18	26	8.8	6.6
1 000	13	11	17	24	8.3	6.2

Table 9: Results from the ATLAS search for a heavy, narrow Higgs boson state decaying to two *Z* bosons, for the vector-boson fuson production mode. Decay modes considered include $\ell\ell\ell\ell\ell$, $\ell\ell\nu\nu$, $\ell\ell qq$, and $\nu\nu qq$. Tabulated as a function of m_H are the observed 95% CL limits on $\sigma \times BR(H \to ZZ)$, the expected limits, and the 1- and 2- σ ranges around the expected limits.



Figure 26: 95% CL exclusion contours in the 2HDM (a) Type-I and (b) Type-II models for $\cos(\beta - \alpha) = 0.1$, shown as a function of the parameters $\cos(\beta - \alpha)$ and $\tan \beta$. The red hashed area shows the observed exclusion, with the solid red line denoting the edge of the excluded region. The dashed blue line represents the expected exclusion contour and the shaded bands the 1- and 2- σ uncertainties on the expectation. The grey area masks regions where the width of the boson is greater than 0.5% of m_H .